# Seton Hall University eRepository @ Seton Hall

Seton Hall University Dissertations and Theses (ETDs)

Seton Hall University Dissertations and Theses

2006

# A Study of New Jersey's Assessment and Accountability System for Students with Disabilities and its Impact on Special Education Services

Julie Moul Gardner Seton Hall University

Follow this and additional works at: https://scholarship.shu.edu/dissertations Part of the <u>Accessibility Commons</u>, <u>Educational Assessment</u>, <u>Evaluation</u>, and <u>Research</u> <u>Commons</u>, and the <u>Special Education and Teaching Commons</u>

#### **Recommended** Citation

Gardner, Julie Moul, "A Study of New Jersey's Assessment and Accountability System for Students with Disabilities and its Impact on Special Education Services" (2006). Seton Hall University Dissertations and Theses (ETDs). 2348. https://scholarship.shu.edu/dissertations/2348

# A STUDY OF NEW JERSEY'S ASSESSMENT AND ACCOUNTABILITY SYSTEM FOR STUDENTS WITH DISABILITIES AND ITS IMPACT ON SPECIAL EDUCATION SERVICES

#### BY

#### JULIE MOUL GARDNER

# Dissertation Committee

Barbara Strobert, Ed.D., Mentor Michael Osnato, Ed.D. Janet L. Parmelee, Ed.D, J.D. Lloyd M. Leschuk, Ed.D.

Submitted in partial fulfillment of the requirements for the Degree of Doctor of Education Seton Hall University 2006

#### ABSTRACT

A Study of New Jersey's Assessment and Accountability System for Students with Disabilities and its Impact on Special Education Services

Use of high stakes assessment for special needs students is a key element in the current educational reform movement. State assessment systems use these tests to determine and report school accountability scores, distribute school funding, and assess school rewards and sanctions. The impact of including these students' test score on the district accountability reports has not yet been determined. Directors of special services, the primary individuals for supervising or coordinating district special education programs, have an extensive understanding of administrative and functional knowledge of special education services and are therefore able to provide insight concerning the impact of accountability on special education services.

This study collects and describes directors' of special services perceptions on the impact of high-stakes assessment and special education services. Data were gathered in this study through both survey and focus group format.

The findings of this study indicate accountability in special education has improved the instructional process for students with disabilities. Professional development opportunities have improved for general education teachers. The inclusion of students with disabilities is viewed with mixed results. We still do not know the full impact of including students in high-stakes large-scale testing. Systematic concerns related to the reliability and validity of the alternate assessment process exist. Questions remain on whether the assessment meets the desired outcome of measuring student performance.

TADIE OF CONTENTS
TABLE OF CONTENTS
PDro 4 month
EDICATION
CENOWI EDGEMENTS
CKINOW LEDGEMEINTS
(a) the subscription of the second se Second second secon second second sec
ατ ΛΕΤΑΡΙΕς
μαστέο 1. πιτρωδιιστίων
Statement of the Drohlem
Dumore of the Study
Significance of the Study
Pagagrah Quagtions
Drosedures
Limitations of the S tu d y
Definition of Terms
S uninn ary
LIADTED 11+1 TEDATIDE DEV/LEW/
HAPTER II: LITERATURE REVIEW
HAPTER II: LITERATURE REVIEW Overview
HAPTER II: LITERATURE REVIEW Overview Legislation Standards Based Beform
HAPTER II: LITERATURE REVIEW Overview Legislation Standards Based Reform
HAPTER II: LITERATURE REVIEW Overview Legislation Standards Based Reform Adequate Yearly Progress
HAPTER II: LITERATURE REVIEW Overview Legislation Standards Based Reform Adequate Yearly Progress Accountability Through Assessment
HAPTER II: LITERATURE REVIEW Overview Legislation Standards Based Reform Adequate Yearly Progress Accountability Through Assessment Assessment Accommodations
HAPTER II: LITERATURE REVIEW Overview Legislation Standards Based Reform Adequate Yearly Progress Accountability Through Assessment Assessment Accommodations Alternate Assessment
HAPTER II: LITERATURE REVIEW Overview Legislation Standards Based Reform Adequate Yearly Progress Accountability Through Assessment Assessment Accommodations Alternate Assessment Assessment Validity and Reliability
HAPTER II: LITERATURE REVIEW   Overview   Legislation
HAPTER II: LITERATURE REVIEW   Overview   Legislation   Standards Based Reform   Adequate Yearly Progress   Accountability Through Assessment   Assessment Accommodations   Alternate Assessment   Assessment Validity and Reliability   New Jersey Assessment and Accountability System   Effects of Inclusion in Testing
HAPTER II: LITERATURE REVIEW Overview Legislation Standards Based Reform Adequate Yearly Progress Accountability Through Assessment Assessment Accommodations Alternate Assessment Assessment Validity and Reliability. New Jersey Assessment and Accountability System Effects of Inclusion in Testing Summary.
HAPTER II: LITERATURE REVIEW Overview Legislation Standards Based Reform Adequate Yearly Progress Accountability Through Assessment Assessment Accommodations Alternate Assessment Assessment Validity and Reliability New Jersey Assessment and Accountability System Effects of Inclusion in Testing Summary
HAPTER II: LITERATURE REVIEW Overview Legislation Standards Based Reform Adequate Yearly Progress Accountability Through Assessment Assessment Accommodations Alternate Assessment Assessment Validity and Reliability New Jersey Assessment and Accountability System Effects of Inclusion in Testing Summary
HAPTER II: LITERATURE REVIEW Overview
HAPTER II: LITERATURE REVIEW Overview
HAPTER II: LITERATURE REVIEW Overview Legislation Standards Based Reform Adequate Yearly Progress Accountability Through Assessment Assessment Accommodations Alternate Assessment Assessment Validity and R eliability New Jersey Assessment and Accountability System Effects of Inclusion in Testing S umm ary HAPTER III: METHODOLOGY Overview Methodology Research Design
HAPTER II: LITERATURE REVIEW Overview Legislation Standards Based Reform Adequate Yearly Progress Accountability Through Assessment Assessment Accommodations Alternate Assessment Assessment Validity and Reliability New Jersey Assessment and Accountability System Effects of Inclusion in Testing Summ ary HAPTER III: METHODOLOGY Overview Methodology Research Design Subjects
HAPTER II: LITERATURE REVIEW Overview Legislation Standards Based Reform Adequate Yearly Progress Accountability Through Assessment Assessment Accommodations Alternate Assessment Assessment Validity and Reliability New Jersey Assessment and Accountability System Effects of Inclusion in Testing Summ ary HAPTER III: METHODOLOGY Overview Methodology Research Design Sub jects PASE Survey Participants
HAPTER II: LITERATURE REVIEW Overview
HAPTER II: LITERATURE REVIEW Overview Legislation Standards Based Reform Adequate Yearly Progress Accountability Through Assessment Assessment Accommodations Asternate Assessment Assessment Validity and Reliability New Jersey Assessment and Accountability System Effects of Inclusion in Testing Summ ary HAPTER III: METHODOLOGY Overview Methodology Research Design Subjects PASE Survey Participants Focus Group Participants Data Collection
HAPTER II: LITERATURE REVIEW Overview Legislation Standards Based Reform Adequate Yearly Progress Accountability Through Assessment Assessment Accommodations Assessment Accommodations Alternate Assessment Assessment Validity and Reliability New Jersey Assessment and Accountability System Effects of Inclusion in Testing Summ ary HAPTER III: METHODOLOGY Overview Methodology Research Design Subjects PASE Survey Participants Focus Group Participants Data Collection PASE Survey
HAPTER II: LITERATURE REVIEW Overview Legislation Standards Based Reform Adequate Yearly Progress Accountability Through Assessment Assessment Accommodations Alternate Assessment Assessment Validity and R eliability New Jersey Assessment and Accountability S ystem Effects of Inclusion in Testing S umm ary HAPTER III: METHODOLOGY Overview Methodology Research Design S u b jects PASE Survey Participants Focus Group Participants Data Collection PASE Survey Focus Group
HAPTER II: LITERATURE REVIEW O v ervie w Legislation Standards Based Reform Adequate Yearly Progress Accountability Through Assessment Assessment Accommodations Alternate Assessment Assessment Validity and R eliability New Jersey Assessment and Accountability System Effects of Inclusion in Testing S umm ary HAPTER III: METHODOLOGY Overview Methodology Research Design Su b jects PASE Survey Participants Focus Group Participants Data Collection PASE Survey Focus Group Instrument Design

The second s
Focus Group Design
Data Analysis
Analysis of PASE S urv e y
Analysis of the Focus Group
Summary
HAPTER IV: RESEARCH FINDINGS
Overview
Quantitative Research Results
Research Question 1
Research Question 2
Hypothesis for Question 2
Research Question 3
Hypothesis for Question 3
Research Question 4
Hypothesis for Question 4
Research Question 5
Research Question 6
Research Question 7
Research Question 8
Summary of Quantitative Research Results
Qualitative Research R e s u l t s
Summary
HAPTER V. SUMMARY
Summary
Findings
Implications
Recommendations
Future Research
State and National Policy Implications
Summary
Summary
Leferences
Appendix A: Survey Instrument
Annendix B: Foous Group Instrument
ppondix B. 10000 0100p m3 (10m on (

# DEDICATION

This dissertation is dedicated to my husband Tom and my children Ian and Caitlyn. I cannot express enough gratitude to each of them for their personal support and sacrifices during this study.

It is also dedicated to my parents, James and JoAnn Moul for instilling in me the desire to learn and achieve. They have been responsible for giving me the necessary skills to be successful in life.

Lastly, I dedicate this dissertation to the Morris-Union Jointure Commission's superintendent, assistant superintendent and my fellow principals for providing the incentive to pursue my doctoral degree in education administration. It is their commitment to the field of education and special education that inspired me to continue my professional growth.

#### ACKNOWLEDGEMENTS

Grateful acknowledgement is expressed to the members of my dissertation committee at Seton Hall University. Dr. Barbara Strobert, my mentor professor, provided professional guidance throughout the project. To Dr. Michael Osnato, much appreciation is expressed for his review of the details of the drafts of this research project. To Dr. Lloyd Leschuk for providing me with valuable feedback and to Dr. Janet Parmelee, my sincerest appreciation who brought important educational perspectives as well as the grammatical corrections throughout this process. My thanks to each of my committee members for giving freely of their time and intellect to help me complete this project.

I truly appreciate all of the directors of special services support in this process and the time they took away from their busy jobs to complete the survey and participate in the focus group.

# LIST OF TABLES

1 Summary of Frequencies for Changes in Student Achievement and Instruction	75
2 Summary of Received Surveys with Respect to DFG	77
3 Descriptive Statistics for Accountability by DFG Group	77
4 Test of Homogeneity of Variances for Accountability by DFG Group	78
5 ANOVA for Accountability by DFG Group	78
6 Summary of Received Surveys with Respect to District Enrollment Groups	79
7 Descriptive Statistics for Accountability by District Enrollment Group	80
8 Test of Homogeneity of Variances for District Enrollment Group	81
9 ANOVA for District Enrollment Group	81
10 Summary of Received Surveys with Respect to Years of Experience	82
11 Descriptive Statistics for Accountability by Years of Experience	83
12 Test of Homogeneity of Variances for Years of Experience	84
13 ANOVA of Years of Experience	84
14 Summary of Frequencies with Respect to Changes in Use of Testing Accommodations	85
15 Summary of Frequencies with Respect to Professional Development	86
16 Summary of Frequencies with Respect to the Effectiveness of the Alternate Proficiency Assessment	86
17 Summary of Frequencies with Respect to the Effectiveness of Inclusion in Testing	87
18 Summary of Focus Group Responses	92

1 Chapter I

#### Introduction

The purpose of this study was to explore the consequences of large-scale high-stakes assessment, with specific reference to the New Jersey Alternate Proficiency Assessment results for students with disabilities through the point of view, experiences, and perceptions of the directors of special services in 100 public schools in central and northern New Jersey. The information gathered through the use of a survey provided insights into consequential aspects of large-scale assessment as it is practiced in New Jersey.

Educators, administrators, parents, citizens, and policymakers need to know the extent to which all students are learning and benefiting from the educational programming in local districts. As a means to assess this progress, states have implemented large-scale, high-stakes testing. Improvement in test scores is the primary measure that states use to report to the public whether schools are improving and students are achieving. In the political arena, accountability in education equates to high-stakes testing and disclosure of the test results to the public.

Accountability has become linked to financial assistance, school quality, teacher competency and student progress. Numerous researchers view assessment as a process to motivate teachers to provide quality instruction, motivate students to achieve, measure a student's performance, and ensure quality educational services (Browder, Fallin, Davis, & Karvonen (2003), Kampfer, Horvath, Kleinert, & Kearns (2001), Thompson, Quenemoen, Thurlow, Ysseldyke (2003), Thompson, & Thurlow, (2001), Turner, Baldwin, Kleinert, & Kearns (2001). However, conflicting evidence is emerging that questions the impact of high-stakes testing, especially with regard to improved student learning (Amerein & Berliner, 2002). Additional questions have focused on the effects on the curriculum, student and teacher attitudes, school climate, and the financial impact of testing. (Elliott, Erickson, Thurlow, & Shriner, 2000).

No Child Left Behind Act of 2001 (NCLB 2001) and the quest for improved schools have elevated the importance of assessment results for the purpose of school improvement and student achievement for all students, including students with disabilities. Schools can no longer ignore the challenges faced by students with disabilities and exclude them from standardized statewide and district-wide testing in reading, math and science.

State and district reports must aggregate the data as well as disaggregate the data by economic, ethnic, disability, gender and English-proficiency status (NCLB 2001, Title I, Part A, Section 1111a2C, 2001). According to the U.S. Department of Education (Facts Page on Adequate Yearly Progress, 2003), the purpose of disaggregating the data is to assure that "failure cannot hide." The goal of NCLB 2001 is to close the achievement gaps between students of different groups by, "holding schools accountable for the achievement of all subgroups," thereby assuring that "no child is left behind" (U.S. Department of Education, Facts Page on Adequate Yearly Progress, 2003). Because these results are readily available in state reports, the public, media, policymakers, and educational researchers are able to use the results to interpret performance levels and performance trends for students with disabilities.

While the idea of students with disabilities being included in a standards-based reform movement is welcomed by educators and parents, it remains unclear how to accurately and appropriately assess special education students, how to assure the students can achieve the same educational standards as other students and the ultimate impact of reporting test results of special education students on both the schools' and districts' adequate yearly progress indices (Albrecht & Joles (2003); Cohen & Heumann, (2001).

#### Statement of the Problem

This study will employ quantitative and qualitative research to explore the method consequences of large-scale high-stakes assessment and its impact on the increased demand for accountability on special education services in 100 public school districts in central and northern New Jersey.

With the passage of NCLB 2001, educators, for the first time, must confront what it really means for all students to achieve high standards, including students with the most severe disabilities. Students with disabilities have historically been exempt from highstakes testing with limited expectations of accountability for performance. This legislation indicates a policy shift away from a special education system grounded in meeting individualized goals to inclusion in a standardized system of accountability. The focus in special education has changed from concerns regarding inclusion of students into regular education classrooms to inclusion of these students in the standardized assessment process.

With this tremendous task facing educators, a central question remains how to accurately and appropriately assess this diverse group and the impact of reporting the

results on a school district's adequate yearly progress. Coupled with this diversity is the increasing number and percentage of students receiving special education services both in New Jersey and nationally. According to the *Quality Counts 2004 Report* (Education Week, 2004), special education services are increasing at a rate six times faster than the regular education population. The combination of an increasing number of special education students and the stringent accountability requirements dictates a need to explore the consequences of large-scale high-stakes assessment for students with disabilities.

#### Purpose of the Study

The purpose of this study was to explore the consequences of large-scale high-stakes assessment, with specific reference to the New Jersey Alternate Proficiency Assessment results for students with disabilities through the point of view, experiences, and perceptions of the directors of special services in 100 public schools in central and northern New Jersey. The information gathered through use of a survey will provide insights into consequential aspects of large-scale assessment as it is practiced in New Jersey. Questions will focus on student performance, accommodations, alternate assessments, accountability, and current and emerging practices and issues. The focus group venue is designed to provide insight and experiences from the participants into the consequential aspects of large-scale assessment and the impact on accountability of special education services as it is practiced in New Jersey.

# Significance of the Study

The inclusion of students with disabilities in large-scale high-stakes testing is mandated by federal and state legislation. As reported by the National Center for Educational Outcomes (Thurlow, Wiley & Bielinski, 2003), special education advocates support the inclusion in the assessment process based on the ideal that the information provided by testing will result in improved educational opportunities for students with disabilities. This trend holds both promise and peril for these students. States have been making progress to include performance scores of special education students in the district adequate yearly progress as required by NCLB 2001 According to Education Week *Quality Counts 2004* report (2004, p.7), 43 states and the District of Columbia include information on the test participation rates and performance of students with disabilities in the reporting system.

The Council for Exceptional Children (Allbritten, Mainzer, & Zeigler 2004) reports that despite agreement that students with disabilities should be included in testing and receive the benefits of the test outcomes, questions concerning the reliability and validity of test results when accommodations and alternate assessments are used, the accuracy of reporting those results, and ultimately the impact on special education services of including the results in the state's report card remain.

Individuals with Disabilities Act, 2004 and NCLB 2001, require students with disabilities to receive appropriate accommodations and modifications for standardized assessments as determined by the individualized education program (IEP) team. However, the use and types of accommodations are controversial and raise questions relating to the validity of the standardized assessment when accommodations are allowed.

The IEP team is also given the responsibility to determine if a student should be assessed through the use of an alternative assessment, when even with accommodations, the student is unable to take the standardized test. Prior to the regulation change on December 9, 2003 (U.S. Department of Education, 2003), districts were required to report all scores, even scores of assessments taken by students with the most significant cognitive disabilities, based on the same grade-level standards. These scores were frequently scored as non-proficient, thus hindering state and district efforts to reach adequate yearly progress as defined in NCLB 2001.

Under the final rules published in the December 9, 2003 Federal Register, states are permitted to establish alternative assessment standards for students with the most significant cognitive disabilities for the purpose of satisfying the adequate yearly progress requirement of NCLB 2001. The alternate academic achievement standards must clearly define the connection between the instructional content appropriate for non-disabled students and the related knowledge and skills that may serve as a basis for definition of proficient achievement for students with the most significant cognitive disabilities (Special Education Report, 2004, p. 6).

States are permitted to use alternate achievement standards to evaluate the performance of students with the most significant cognitive disabilities and to give equal weight to proficient and advanced performance based on the alternate standards in calculating adequate yearly progress, provided the number of proficient and advanced scores based on the alternate achievement standards does not exceed 1 percent of all students in the grades tested at the state or district level. (Special Education Report, 2004, p. 7).

Even with these changes in NCLB 2001, states may not exclude the scores of students who exceed the percentage cap from the adequate yearly progress calculations. States must count the non-proficient scores for these students and for whom no exception is

granted (Special Education Report, 2004, p. 8). There is no limit to the number of students who take an alternate assessment, just the percentage of scores based on alternate assessment standards that can be counted as proficient when calculating adequate yearly progress.

Although states have some flexibility in reporting the scores of students with the most significant cognitive disabilities, districts are still required to include these students' scores as well as all students with disabilities scores on school and district report cards. This change makes an allowance, and thus an explanation for minimal progress, for the reporting of the scores for students with the most significant cognitive disabilities. However, the impact of the scores for students not meeting the 1% allowance on district adequate yearly progress is still unresolved. Even though the federal government has addressed the reporting of scores, questions remain as to the validity of the alternate assessments and the accommodations permitted when testing students that do not fall within the 1% exception. Therefore, it seems vital to explore the consequences of the use of accommodations and alternate assessments on large-scale high-stakes test results for students with disabilities and the impact of reporting these results on special education services.

#### Research Questions

On the basis of the previous research and related literature, the research questions were designed to provided insights into consequential aspects of large-scale assessment as it is practiced in New Jersey by answering the following questions:

1. What is the perception of the directors of special services toward changes in student achievement and instruction of special education students since the inclusion of students with disabilities in the New Jersey state assessment system?

2. What is the perception of accountability in special education of directors of special services in low socioeconomic districts (DFG Group) as compared to the perception of accountability in special education of directors of special services in high socio-economic districts (DFG Group)?

3. What is the perception of accountability in special education of director's of special services in high-enrollment districts as compared to the perception of accountability in special education of directors of special services in low-enrollment districts?

4. What is the perception of accountability in special education of directors of special services with less experience as compared to the perception of accountability in special education of directors of special services with more experience?

5. What is the of perception of the directors of special services toward changes in use of testing accommodations since the inclusion of students with disabilities in the New Jersey State assessment system?

6. What is the diversity of perceptions of the directors of special services toward changes in professional development since the inclusion of students with disabilities in the New Jersey State assessment system?

7. What is the diversity of perceptions of the directors of special services toward the effectiveness of the Alternate Proficiency Assessment since the implementation of New Jersey's State alternative assessment system for students with severe disabilities?

8. What is the diversity of perceptions of the directors of special services toward the effect of including students with disabilities in the New Jersey statewide assessment program?

#### Procedures

This descriptive study employed two methodologies: administering a survey Perceptions of Accountability in Special Education (quantitative design) and conducting a focus group session (qualitative design), in an attempt to explore directors of special services' perceptions regarding the consequences of high-stakes assessment for students with disabilities and the impact of accountability on special education services.

Data were assembled from the survey and the focus group session for analysis. Quantitative analysis procedures included basic descriptive statistics and an analysis of variance (ANOVA) to examine differences between the participant groups involved in the study. Qualitative analysis included coding of the director of special services comments according to themes and patterns.

#### Cint billing.

and institutes in populations, subspace, and policies of the two source resource and institute lifest forces any leafs to different furthings with organic to the

## Limitations of the Study

This investigation was limited by the following elements:

1. New Jersey's assessment and accountability is unique compared with other states. Information collected from this study may not be generalized to other large-scale assessments.

2. Participants' responses were representative of their individual experiences and past and current job responsibilities.

3. This study was not designed to show a causal relationship between the consequences of large-scale testing and educational outcomes for students with disabilities.

4. Differences in populations, practices, and policies in the 100 school districts in central and northern New Jersey may lead to different findings with regard to the questions addressed in this study.

5. Abbott districts with an enrollment of greater than 9000 were not included in the study. This may lead to different results with regard to the DFG socioeconomic group findings.

# De finition of Terms

The following selected terms are defined as they were used throughout this study:

Accommodations: Changes made to the assessment procedures in order to provide a student access to information and an equal opportunity to demonstrate knowledge and skills. Accommodations should be chosen based on the IEP team's recommendations and reflect accommodations currently used in daily instruction.

Accountability: Includes schools and districts being accountable for student outcomes by reviewing, evaluating, and improvement in student achievement. Due to NCLB 2001, accountability is linked to rewards and sanctions.

Abbott District: Refers to the New Jersey school districts that receive state aid that is calculated to provide them with the same per-pupil operating budget as would be found in New Jersey's wealthiest school districts. "Abbott" is the short-hand description of a series of New Jersey Supreme Court decisions growing out of litigation filed in 1981 on behalf of children residing in New Jersey's most economically disadvantaged municipalities.

Adequate Yearly Progress (AYP): Refers to the growth rate in the percentage of students who achieve the states definition of academic proficiency. Each state will set the AYP gains every school must meet to reach 100%proficiency at the end of 12 years. Schools are held accountable for the achievement of all students, not just average student performance.

Aggregation: The accumulation of a like data into a larger set. For example, the computation of individual school test scores into a district's score.

<u>Alternate assessment: A method of measuring the performance of students who are</u> unable to participate in statewide assessments, even with accommodations.

Core Curriculum Content Standards: New Jersey's descriptions of the knowledge and skills students should acquire in a particular subject area. New Jersey includes three content areas in the state-wide assessment process: language arts, mathematics, and science

<u>Criterion Referenced Tests</u>: A test that allows its users to make score interpretations in relation to a functional performance level.

Disaggregate: Test results are analyzed by economic, ethnic, disability, gender, and English-proficiency status.

Grade Eight Proficiency Assessment (GEPA): New Jersey eighth grade assessment to determine whether students are making sufficient progress in mastering the knowledge and skills they will need to pass the 11th grade assessment.

High School Proficiency Assessment (HSPA): New Jersey 11th grade assessment used to determine student achievement in reading, writing, and mathematics as specified by the New Jersey Core Curriculum Content Standards.

High-stakes Assessment: An assessment system for which the results have important consequences for students, teachers, and school districts. Examples of high-stakes include promotion, certification, graduation, or school and district rewards or sanctions.

IDEA 2004: An amendment to the Individuals with Disabilities Act which mandates that all students, including students with disabilities, be included in state and district-wide assessments or receive alternate assessments.

peribilis accessions designed to moreore program inserti soldaring Pine Access's data of configured standards for finae statemic toffi scores dissibilities with see totable to periodynas in the MULTER, URIGN, FERM

Individualized Education Plan (IEP): The individual program plan for the delivery of services to each student with a disability. The intent of the IEP is to assure the student receives an appropriate education in a least restrictive environment.

Large-scale Assessment: Assessments for purposes outside the classroom. Largescale assessments are usually standardized and give to large groups of students at the same time. Purposes for large-scale assessments include accountability, program evaluation, selection of students for special programs.

<u>Modifications: Changes</u> made to the assessment procedures in order to allow a student to participate. Modifications used during statewide testing should be consistent with instructional assessment procedures used in the classroom.

New Jersey Alternative Proficiency Assessment (NJAPA): Alternate assessment is a portfolio assessment designed to measure progress toward achieving New Jersey's state educational standards for those students with severe disabilities who are unable to participate in the NJASK, GEPA, HSPA

New Jersey Assessment of Skills and Knowledge (NJASK): New Jersey comprehensive, multi-grade assessment program for elementary school age students. The results of the assessment are intended to identify students who need additional instructional support in order to reach the state core curriculum content standards. Replaced the ESPA assessment.

Out-of-level Testing: Administering a test that is designed primarily for people of an age or grade level above or below that of the test taker.

Standards-based Reform: Educational reform in which curriculum is based on content that is clearly defined and meets explicit levels of performance.

Special Review Assessment (SRA): New Jersey alternate assessment for high school students that provides the student with the opportunity to exhibit their understanding and mastery of the HSPA skills in contexts that are familiar and related to their experiences.

# Summar y

Use of high-stakes assessment for special needs students is a key element in the current educational reform movement. State assessment systems use these tests to determine and report school accountability scores, distribute school funding, and assess school rewards and sanctions. In response to IDEA 2004 and NCLB 2001 student with disabilities are no longer exempt from participation in the high-stakes assessments. However, the impact of including these students' test score on the district accountability reports has not yet been determined. Directors of special services, the primary individuals responsible for supervising or coordinating district special education programs, have an extensive understanding of administrative and functional knowledge of special education services and are therefore able to provide insight into the consequences of high-stakes assessment for students with disabilities, and the impact of accountability on special education services. This study collects and describes directors' of special services perceptions of the impact of high-stakes assessment and special education services. Chapter 2 presents a review of the current review of the literature of high-stakes assessment for students with disabilities. The chapter discusses governmental and legislative impacts on accountability and assessments in special education, standards-based reform, accountability and adequate yearly progress, and alternate assessment and testing accommodations. Chapter 3 describes the setting in

which the research takes place, the data collection process through use of the Perceptions of Accountability in Special Education (PASE) survey (quantitative design) and the focus group (qualitative) and the methodologies that were employed. Chapter 4 summarizes the results of the data analysis for the survey questions and the focus group responses to find out about the directors' of special services perceptions on the impact of high-stakes assessment and special education services. Chapter 5 discusses the research findings and the implications for future research and policy initiatives in the realm of special education services.

# Chapter II

## Literature Review

#### Overview

This chapter reviews the literature relating to the topic of assessment and its impact on the increased demand for accountability on special education services. Among the topics discussed in this chapter are: governmental and legislative impacts on accountability and assessments in special education, standards-based reform, accountability and adequate yearly progress, and alternate assessment and testing accommodations.

#### Legislation

Elementary and Secondary Education Act (ESEA) was reauthorized in 1994 as the Improving America's Schools Act (IASA). IASA required that in Title I funded schools, disadvantaged children served by Title I would meet the same challenging standards for student achievement as expected of other children. All students in the school are held to the same standards, and the progress of all students is measured by these assessments and reported to the public. Based on the assessment reports, schools would make instructional and structural changes needed so the expectations of all students are raised providing opportunities to work toward high standards.

The IASA, the Individuals with Disabilities Education Act (IDEA '97), IDEA 2004 and the No Child Left Behind Act (NCLB 2001 2001) are the most recent statutes to impact the educational reform movement that considers the learning needs of all students,

including those with disabilities. Each of these laws has the common requirement that students with disabilities be included in the assessment system.

IDEA '97 first identified an alternate assessment option for students who cannot participate in general assessments with accommodations. The law implicitly defines state and local assessments as contributing to a student's educational opportunities for which access must be granted. The 1997 IDEA Amendments require that states establish performance goals and indicators for students with disabilities and report student progress to the public (34 CFR 300.138).

On December 3, 2004, IDEA 2004 was signed into law. Specific to this legislation is the expansion of the requirements and definitions of assessment and alternate assessments. The new requirements in IDEA 2004 for explicit participation in the general curriculum and inclusive educational assessments are closely tied to the model for inclusion. IDEA 2004 states:

All children with disabilities be included in all general state and districtwide assessment programs, including assessments described under section 1111 of the ESEA of 1965, with appropriate accommodations and alternate assessments where necessary and as indicated in their respective individualized education program (612 [16] [A]).

IDEA 2004 requires states or local education agencies to develop guidelines for the provision of appropriate accommodations (612 [16] [B]) and further clarifies alternate assessments. According to IDEA 612 [16] [C] [i], the state or local education agency has developed or implemented guidelines for the participation of children with disabilities in alternate assessments. IDEA 2004 includes specific requirements for alternate assessments in that the assessments are aligned with the state's challenging academic

content standards and challenging student academic achievement standards (612 [16] [C] [ii] [I]).

NCLB 2001 requires that no less than 95% of all students in a school participate in state assessments (1111[2]iii). The requirement is broken down into subgroups, with the requirement that 95% of each subgroup must be tested. Under the Act's accountability provisions, states must describe how they will close the achievement gap and make sure all students, including those who are disadvantaged, achieve academic proficiency. They must produce annual state and school district report cards that inform parents and communities about state and school progress. Reporting annual achievement of all students with regard to academic standards enables teachers and parents to identify areas of need and refine instructional programs. Therefore, assuring all students are given the opportunity to receive instruction related to the state's standard performance standards. Schools that do not make progress must provide supplemental services, such as free tutoring or after-school assistance, take corrective actions and, if still not making adequate yearly progress after 5 years, make dramatic changes to the way the school is run (U.S.D.O.E.).

There is no mandate specifying the number or percentage of students with disabilities in states who must participate in standard state assessments. The United States Department of Education (USDOE) (Heumann & Warlick, 2000, p. 8) states, "It is expected that the number of students participating in alternate assessments will be relatively small." NCLB 2001 allows the use of alternate assessments for students with the most severe cognitive disabilities, provided that does not exceed 1% of all students assessed and that the alternate assessment process is aligned with the state's academic content standards. The assessment must reflect professional judgment of the highest learning possible for those students. The majority of students with disabilities should be participating in standard statewide assessments with appropriate accommodations as needed.

#### Standards-based Reform

Accountability is a critical aspect of standards-based reform. The rationale for accountability systems is the belief that education can be improved when clear standards for student achievement are communicated to students and educators, achievement toward those standards is measured, and appropriate consequences are linked to levels of student achievement (Lehr & Thurlow, 2003).

Standards-based reform promotes the setting of standards, identifying indicators of how to successfully meet those standards, and ways to measure student progress toward these indicators. The emphasis of standards-based reform is on students achieving high academic standards and that the achievement is documented through testing and assessment (Thurlow, et. al., 2003). The standards-based educational reform movement is intended to produce improvements in instruction and curriculum. Thompson et. al. (2001) defines standards-based reform as including these assumptions:

- 1. All children are expected to work toward state or district high standards;
- 2. States and districts measure all children's progress toward the standards.
- 3. Schools use measurement data to make improvements in curriculum and

Utmahunda-bened reform permittees the setting of summinda, identifying indicators of 19

considered instructional constants. This standards one skills and oblight marked by all stations including the dovelopment of high content area knowledge and understand the shifting to base in Kin wissing to now and Stifts out settings, and the shifting to apply

20

insure action to allow all students is to succeed; and

4. States, districts, and schools are held accountable for student success.

Accountability through high-stakes assessment has become a vital component of the educational process with rewards and sanctions for school improvement at the state and local level.

As part of the standards-based reform effort inherent in NCLB 2001, states have developed educational standards along with statewide assessment systems that measure student progress in the core content areas and serve as a basis for district accountability. Content standards are educational standards that define what students are expected to know and to be able to do. They do not guide day to day instruction and are not considered instructional curricula. The standards are skills and abilities needed by all students including the development of basic content area knowledge and understanding, the ability to transfer knowledge to new and different settings, and the ability to apply the learned skills to career or vocational paths, citizenship, or technology (Thompson, et. al, 2001).

Although all students are expected to know and work toward the same high standards, the methods of assessment and the performance expectations among student groups may differ. To assist educators in assessing students appropriately, states have developed performance standards, which are used to describe the quality of the performance expected for proficiency on the content standard. Benchmarks and performance indicators are used to further define measurable and observable skills expected at each grade level. Even though students will vary in their performance levels, they are expected to learn the same high quality content using a wide range of instructional strategies and methods. According to guidance in the IASA on standards, assessments, and accountability, students with disabilities may need modification to ensure appropriate instructional approaches, but generally all students need to be working toward the same challenging standards (IASA, as cited in Thompson, et. al, 2001, p. 20).

States differ on the design of the content standards, performance indicators and progress indicators. To ensure the inclusion of all students, some states have chosen to expand their content standards to include fundamental skills as performance indicators, but maintaining the same standards for all students while others have separate standards or no common standards for students with severe disabilities. Other states offer examples of skills and performance indicators and then invite IEP teams to select a student's performance indicators from the examples (Thompson, et. al, 2001). However, these differences between states do not lend themselves to easy comparison of data among states for accountability purposes.

#### Adequate Yearly Progress

Assessment of student progress is a vital component of educational reform. However, prior to the year 2002 state assessment systems were not tied to sanctions or rewards and were viewed as a process not as a means for improvement.

On January 8, 2002, NCLB 2001 was signed in to law (U.S. Department of Education, 2003, Welcome Letter). With the passage of this legislation, the accountability movement in education reached a new height.

The Education Commission of the States (ECS) defined accountability as, "systematic collection, analysis, and use of information to hold schools, educators, and others responsible for the performance of students and the education system" (ECS, 1999 as cited in Thompson et al. 2003, p. 7). Accountability is simply evidence that schools are doing what they are supposed to be doing.

Federal policy and the quest for improved schools has elevated the importance of assessment results for the purpose of school improvement, student achievement, and accountability. NCLB 2001 legislation contains a comprehensive requirement of accountability for states and local education agencies to report annual achievement of all students with regard to academic standards. Among the accountability requirements of NCLB 2001 is that each state must develop and implement a statewide accountability system that ensures all local school districts, and public elementary and secondary schools, make adequate yearly progress (AYP) on annual state assessments in reading and math with science being added by 2005. Adequate yearly progress applies the same high standards of academic performance to all public school students and measures progress of the public schools and local districts on state academic assessments. Standards must be defined by states to include measurable annual goals and objectives for all students and for subgroups, including students with disabilities. Adequate yearly progress must include a timeline for ensuring that each group of students meets or

exceeds a state determined "proficient" level of performance on the state assessment. NCLB 2001 requires states to:

provide for (I) the participation in such assessments of all students; and (II) the reasonable adaptations and accommodations for students with disabilities as defined under section 602(3) of IDEA '97 as necessary to measure the academic achievement of such students relative to State academic content and State student academic achievement standards." (1111[b]3six)

States must provide assessment reports that teachers and parents can use to identify student needs and achievement in relation to state academic standards (NCLB 2001). The assessments must also, "enable results to be disaggregated within each state, local education agency and school by students with disabilities as compared to students without disabilities" (1111[b]3). Assessment is viewed as a process to motivate teachers to provide quality instruction, motivate students to achieve, measure a student's performance, resulting in quality educational services. However, conflicting evidence is emerging that questions the impact of high-stakes testing, especially with regard to improved student learning (Amerein & Berliner, 2002). Additional questions have focused on the effects on the curriculum, student and teacher attitudes, school climate and the financial impact of testing.

Sections 1111(b)(2)(G) and (I) of NCLB 2001 state that schools must show each demographic group is making progress in meeting state objectives and that at least 95 % of the students in each group took the state assessment. According to Section 1111(b)(2)(c)(v), schools do not have to disaggregate data for a demographic group if the number of students is too small to yield statistically reliable information, or the results would reveal personally identifiable information about an individual student.

#### Accountability Through Assessment

Prior to the implementation of NCLB 2001, students with disabilities were not required to participate in state assessments if their IEP exempt them from the state assessment process. The IDEA 2004 Amendments required participation by all students with disabilities and for states to report the performance of these students at the same rate as for students without disabilities. As a result of NCLB 2001 and IDEA 2004 legislation, states and local education agencies are now held accountable for assessing these students and reporting the test results.

States and local education agencies developed measurements for systems accountability and student accountability to meet the federal mandates. System accountability is designed to improve educational programs, whereas student accountability is designed to motivate students to do their best. Both types of accountability can have unintended negative consequences as well as the intended positive consequences. Thurlow, Elliot, Ysseldyke (1999) state that "A system is accountable for all students when it makes sure that all students count in the evaluation program of the education systems." NCLB 2001 requires that:

Each state plan shall demonstrate that the State has developed and is implementing a single, statewide State accountability system that will be effective in ensuring that all local education agencies, public elementary schools, and public secondary schools make adequate yearly progress as defined under this paragraph (1111[b]1).

As the general education program has moved from examining content and process to measuring educational results, such as scores on state-wide assessments, so must the evaluation of programs for students with disabilities. Thurlow, Elliott, & Ysseldyke, (1999) list five reasons for including students with disabilities in accountability systems.

- 1. For an accurate picture of education;
  - 2. For students with disabilities to benefit from reforms;
  - 3. To make accurate comparisons;
  - 4. To avoid unintended consequences of exclusion; and
  - 5. To meet legal requirements

McDonnell et al. (1997) list two goals in the efforts to include students with disabilities in assessment and accountability systems. One is to improve the quality of educational opportunities afforded students with disabilities. The second is that inclusion in the assessment process will provide meaningful and useful information about performance of students with disabilities and about the schools that educate them.

The most basic and visible form of accountability is the public reporting of student performance through assessment (EPRRI's Policy Update, 2002). Assessment as defined by Salvia & Ysseldyke (2001) refers to the collection of data thus providing the evidence for accountability. Therefore, it becomes imperative to examine whether current state assessments measure students with and without disabilities equally. State assessments are designed to determine if students have met identified standards of performance related to state content standards. In order to apply the same standard of performance to students with disabilities, the criterion to measure performance must be equally valid for both groups. Since test items are often linked to specific skills from a state's content

26 standards, test items must measure those skills in the same way for all students in order to make assumptions about student performance based on broad standards.

One of the greatest challenges for states has been the inclusion of more students with disabilities in state testing and accountability systems. Previous to the NCLB 2001 mandate variation of the rates of exclusion of students with disabilities ranged from 33% to 87% (Thurlow, et al., 2003, p. 5). As a result of NCLB 2001, Education Week Quality Counts 2004 report shows 13 of 37 states provide participation rates for students with disabilities with 95% or more of the students being tested in reading and math for Grades 4, 8, and 10. Participation rates for students with disabilities ranged from 40 to 100%. Twenty of thirty-one states provided participation rates for general education students with larger proportions of general education students tending to take the tests. Differences in participation rates may reflect the fact that all states do not calculate students the same way when calculating data. All states and the District of Columbia count special education students who take the tests without accommodations, or with accommodations in their participation rates. However, only 26 states count special education students who take state tests with modifications that may result in discrepancies in what is being measured. In the 2002-2003 reporting period, of the 18 states that allow alternative assessments such as portfolio assessment or out of level tests, only 14 states included these results in the reporting of the overall state results (Thurlow & Minnema, 2001).

While more students with disabilities are participating in statewide assessments with and without accommodations, it should be determined whether assessments are benefiting all students the same way. As of 2003, only five states indicated that they

analyzed the assessment results by item for students with disabilities while five states indicated a proposed plan to analyze test items. Fifty percent of the state department of special education directors responded they did not know whether test results were analyzed (Thompson & Thurlow, 2001). These results indicate further discussion between states' assessment agencies and state special education directors is needed to assure test items are assessing student progress or program quality. Eighty percent of state directors of special education reported being involved in their state's development of the definition for adequate yearly progress. (Thompson & Thurlow, 2001)

#### Assessment Accommodations

To ensure testing of all students, states have defined three options for students to participate in the assessment system: participation in the general assessment, typically a large-scale assessment of some type; participation in the general assessment with accommodations to allow students to show what they know; or participation in alternate ways of assessing what students know and are able to do, commonly called alternate assessment.

Students with disabilities may require accommodations and/or modifications in order to participate in the testing process. Section 504 of the Rehabilitation Act of 1973, the Americans with Disabilities Act of 1990 (ADA), IDEA '97, IDEA 2004 and the NCLB 2001 call for accommodations to be provided as necessary to allow students with disabilities to participate in assessments.

Accommodations are alterations in the way a test is administered, without changing the actual test content or performance standard. An assessment accommodation is provided because of a student need, not to give students with disabilities advantage. When students with disabilities use assessment accommodations it is to show what they know without being impeded by their disabilities. Accommodations offset the impact of the disability and are legally required under Section 504 of the Rehabilitation Act, IDEA '97 (Thompson et al., 2001, p. 49). IDEA 2004 requires states or local education agencies to develop guidelines for the provision of appropriate accommodations.

IDEA 2004 states that the state (or in the case of district wide assessment, the local education agency) must develop guidelines for the provision of appropriate accommodations (612 [16] [B]). IDEA 2004 does not distinguish between accommodations and modifications in terms of definition of the appropriateness of their use during the administration of statewide assessments. The use of the terms interchangeably in IDEA has caused confusion in the development of consistent policies for the use of accommodations for standardized assessments. Most states have policies listing acceptable accommodations/modifications allowed during state and district wide testing (Thurlow, Scott & Ysseldyke, 1995); however there is inconsistency among the states' allowances. Thurlow, Seyfarth, Scott & Ysseldyke, (1997) indicate that 39 of the 50 states had policies dictating specific accommodations that would be allowed during testing. Most states allow accommodations but restrict their use on portions of the test.

Modifications are commonly defined as changes in setting, timing, scheduling, presentation or response that are considered to alter what a test is supposed to measure. (Thurlow & Weiner, 2000). Modifications can be in the way assessments are designed or
ontesia melijedne, suj pu gradura nasteroj teroj uznaz salog sobojung, e

(N.J.D.D.E).

administered so that students with disabilities and limited English proficient students can be included in the assessment. Assessment accommodations or adaptations might include Braille forms for blind students or tests in native languages for students whose primary language is other than English (CRESST).

The New Jersey Department of Education defines accommodations and modifications as changes in how a test is presented to a student, or how a student responds on tests, needed in order for a student with a disability to participate in the test (N.J. D.O.E.) Accommodations must be listed in the IEP and Section 504 plans. It is important the accommodations used during statewide testing are consistent with instructional assessment procedures used in the classroom (N.J.D.O.E.). Any accommodations must be recorded in the student's test booklet or the answer sheet. Approved accommodations in the State of New Jersey include: setting, scheduling, test material modifications, and test procedures accommodation and modifications (N.J.D.O.E.).

According to Thurlow and Bolt (2001), 34 states allow reading test items aloud with or without limitations although four of those states limit reading aloud to certain portions of the test. In states where reading aloud is an acceptable accommodation, it is not permitted when the test is intended to measure decoding ability and comprehension but may be permitted when the test is intended to only measure comprehension.

In attempting to meet federal mandates for testing requirements states are caught between conflicting ideologies. The politics of accountability are based on all students achieving at the same high level. Students with disabilities requiring instruction and assessment at levels significantly below that of their grade level peers may limit their meaningful participation in grade level assessments because very few, if any, items on the test would be at an appropriate level of difficulty for these students (Bielinski, Thurlow, Minnema, & Scott, 2000; Gersten & Baker, 2002).

strukturig with disself filler best will we interfere with strukturis without digela these. Update

However, the provision of special education services is based on an understanding of a student's individual difference and needs, thus creating the dilemma for educators. The impact of specific accommodations on a test is a factor in the discrepancies among state policies on acceptable accommodations. Research on accommodations is growing rapidly due to the impact of federal legislation on testing requirements. According to Thurlow & Bolt (2001), accommodations for students with physical or sensory disabilities are routinely approved; however, students with cognitive or behavioral difficulties may not receive the modifications necessary due to a lack of research findings on appropriate testing accommodations. Typical accommodations allowed in the classroom such as using a spellchecker mechanism, expressing the student's knowledge and understanding of a question or concept, or demonstrating listening comprehension are not always permitted as accommodations and modifications in high-stakes testing. Cohen and Huemann (2001) found that providing accommodations will improve the test results of students with disabilities but will not interfere with students without disabilities. Cohen and Huemann clarify the responsibility of the IEP team to exercise authority under IDEA '97 to select individual accommodations and modifications necessary for a student with the disability to participate in state and district assessment. Although state and local education agencies cannot limit the authority of the IEP team to select the accommodations and modifications the results of the test may be deemed invalid under

state or local policies if particular accommodations and modifications are provided but not approved by the state and local agencies as per IDEA 2004.

### Alternate Assessments

IDEA '97 required states to have an alternate assessment in place by July 1, 2000 (612 [17] [A] [ii]. IDEA 2004 further clarifies alternate assessments. According to IDEA 2004 (612 [16] [C] [i], the state or local education agency has developed or implemented guidelines for the participation of children with disabilities in alternate assessments. IDEA 2004 includes specific requirements for alternate assessments in that the assessments are aligned with the state's challenging academic content standards and challenging student academic achievement standards (612 [16] [C] [ii] [I]).

Alternate assessments have evolved over the past several years since they were first required by the 1997 IDEA Amendments. Thompson and Thurlow (2001) report that 75% of the states have at least one alternate assessment option that addresses the needs of students with significant cognitive disabilities. The remaining states have two or more alternate assessment options to address the perceived needs of a range of students in addition to those with significant cognitive disabilities.

Alternate assessments are not a traditional large-scale assessment or an individualized assessment. They are designed to measure performance of students who cannot, even with accommodations, participate in general assessment. Perhaps 1% to 2% of the total population of students typically those with very significant disabilities need alternate assessments (Thompson et.al, 2003, p. 8). Using alternative assessment information,

schools can make broad policy decisions that improve schooling practices assuring all students are successful (Thompson et.al, 2003, p. 9).

As a result of the legislative requirement, states have had to rethink the process used to obtain data on student progress for students with disabilities and how to administer an alternate assessment. Several issues emerged around the "what" of alternate assessment. These issues relate to the alignment of the assessment to the content and performance standards. Students with severe disabilities are often in a curriculum that differs in emphasis from the one that is the course of study for other students. Therefore, the typical test, designed to measure the progress and performance of students in a standard curriculum, often will not be aligned with the curriculum in which students with disabilities receive instruction. Since performance on alternate assessments must be reported with other scores in the assessment system, they must be linked to the high standards that are expected for all students.

Alternate assessments are often viewed as a compromise solution, allowing students to participate in the assessment process while acknowledging their special academic needs. Research-based guidance on successful alternate assessments is limited due to the recent need for an alternate assessment protocol. As a result, the appropriateness of this type of assessment to determine student achievement continues to be reviewed and researched. Alternative approaches are needed to measure the progress of these students toward important educational outcomes. The challenge is that there is a small group of students with severe cognitive deficits or multiple disabilities for whom standard large-scale testing practices and accommodations just do not work. non-new of a for only descated with with stationar with disciplining.

Arguments supporting alternate assessments include the contention that alternate assessments reduce the frustration and random guessing by student who would otherwise be faced with a test that is too difficult. The belief is that the alternate assessment increases the validity of the test scores and the contention that test levels should be matched to the student's IEP goals and objectives (Thurlow, Elliott, & Ysseldyke, 1999). Opponents of alternate assessments argue that the purpose of standardized assessments is to have high expectations for all students and to hold all students to the same standard (Thurlow et al, 1999). There are concerns about the valid interpretation of alternate assessments scores and problems with aggregation of alternate test scores for group reporting purposes. According to Thurlow and Minnema (2001), these issues have not been resolved through empirical research with no definitive data supporting the use or non-use of alternate assessments with students with disabilities.

Ideally, an alternative assessment requires students to actively accomplish complex and significant tasks, while bringing to bear prior knowledge, recent learning, and relevant skills to solve realistic or authentic problems rather than requiring students to choose from a set of provided responses. Exhibitions, investigations, demonstrations, written or oral responses, videotapes, journals, and portfolios are examples of the assessment alternatives we think of when we use the term "alternative assessment."

Research by Quenemon et al (2002) and Thompson & Thurlow (2003) indicates most states commonly use four approaches to alternate assessments. These include portfolio assessment, performance assessment, IEP linked body of evidence, and traditional test formats. States developed assessment criteria based on four assumptions that are the foundation of alternate assessments. These assessments should:  focus on authentic skills and on assessing experiences in community and other real life environments;

measure integrated skills across domains;

2.

3. use continuous documentation methods, if at all possible; and
4. include as critical criteria the extent to which the system
provides the needed supports and adaptations, and trains the student to use them
(Ysseldyke & Olsen,1997).

Thompson and Thurlow (2001) indicate the most common approach to alternate assessment is a portfolio. Student portfolios are a systematic collection of student work that is evaluated and measured against a predetermined scoring criteria. Development of a portfolio assessment for students with the most severe disabilities requires the same thoughtful application analogous to what occurs for general assessment development (Quenemoen, Rigney, & Thurlow, 2002). Portfolios are typically designed so that data could be collected on the educational progress and accomplishments of students with the most severe disabilities. For assessment purposes a portfolio is collection of work, usually drawn from students' classroom work. A portfolio becomes a portfolio assessment when (1) the assessment purpose is defined; (2) criteria or methods are made clear for determining *what* is put into the portfolio, *by whom*, and *when*; and (3) criteria for assessing either the collection or individual pieces of work are identified and used to make judgments about performance. Portfolios can be designed to assess student progress, effort, and/or achievement, and encourage students to reflect on their learning. The results should be designed for use during the school improvement planning process,

to promote access to the general curriculum and to achieve proficiency levels of student performance based on state content standards (Thompson et al., 2001). A portfolio assessment allows students with the most severe disabilities to work toward the same content standards as all students using alternate learning expectations to measure progress.

States have indicated a change in the alternate assessment process since 2001. In 2001, 48% of the states used portfolio assessments as compared to 46% in 2003. The use of checklists or rating scales has increased from 18% in 2001 to 30% in 2003. Other assessments used include IEP analysis: 6% in 2001 and 8% in 2003; local selection of an alternate assessment: 24% in 2001 and 3% in 2003; and assessment under development or revision: 4% in 2001 and 6% in 2003 (Education Week *Quality Counts 2004*).

Eighty percent of state alternate assessments are aligned with the state academic standards content either through grade level, expanded standards, or through a combination of the content areas and functional skills. Four percent of the states conduct alternate assessments that are not aligned to the state standards and 3% allowed IEP teams to select the content used in the alternate assessment process (Thompson & Thurlow, 2001).

The connection to achievement level descriptors between the alternate assessments and the general education assessment has improved since 2001. Sixty-two percent of the states reported using the same descriptors in 2003 as compared to 36% in 2001. New Jersey uses the same descriptors of advanced proficient, proficient, not proficient.

### Assessment Validity and Reliability

Validity refers to the quality of information an assessment yields. It is not an attribute of a test, but an attribute of a specific inference or conclusion based on test scores. If a test is not valid for the purpose used, then it has no inherent value. Traditional measurement research includes three aspects of validity: content validity, criterion-related validity, and construct validity. Construct validity is defined as encompassing the value and social implications of assessment results and is frequently referred to as consequential validity. Content validity refers to how adequately a test samples knowledge and skills that assess the approved written version of the curriculum or content area standards. A test is said to have criterion-related validity if its results parallel some other external criteria. The test results should be similar to the results of other assessments or measures of the same domain.

Large-scale assessment validity and reliability goes through a rigorous review lasting multiple years prior to the implementation of the test. Test items reflect educator and measurement experts understanding of student outcomes and scoring of student responses. Testing companies and state personnel review the test's content specifications, defining the specific skills and knowledge that will be tested and how the students will demonstrate their ability to use the knowledge and skills. Decisions are made on the depth of content coverage and the complexity of the test items. Decisions are made concerning how the students' responses are scored and the scoring criteria. The number of correct and incorrect answers on the test items provides the information needed to identify which students have achieved the knowledge and skills designated by the assessment. Due to years of experience in developing large-scale assessments, the

validity and reliability is rarely questioned or discussed by the public. It is assumed the tests meet the Standards for Educational and Psychological Testing (1999) as developed jointly by the American Educational Research Association (AERA), American Psychological Association (APA), and the National Council on Measurement in Education (NCME).

Previous low expectations for students with severe disabilities have created gaps in knowledge and skills. Until the recent requirement for inclusion of students with disabilities in large-scale testing, educators, measurement experts or test companies had direct experience in defining what successful outcomes are for this group of students at least as these outcomes are defined through measurement of academic achievement (Quenemeon, Thompson, & Thurlow, 2003). Questions have been raised about how to develop assessments and implement the assessments that are appropriate for all students without weakening the content standards and minimizing the rigor of performance (Almond, Quenemoen, Olsen, Thurlow, 2000). Concerns have surfaced as to whether the assessments for these students can be used as an effective measure of student progress based on state content standards and the validity and reliability of these tests.

An assumption of standardized tests is that state assessments measure knowledge and skill within content area in the same way for both students with disabilities and nondisabled students. This assumption has yet to be explored through empirical research (McDonnell et al. (1997). Before making assumptions about the achievement of content standards by students with disabilities based on statewide assessments results, examination to determine if those assessments are valid and reliable for these students is necessary. Since alternate assessments are a more recent development than general assessments, the assumptions underlying them continue to be discussed and researched.

Most states currently do not have data on technical quality of the assessments due to the newness of the alternate assessment process. Researchers have questioned the reliability and validity of alternate assessments. These studies suggest that the concepts of reliability and validity do not have precisely the same meanings for portfolio assessments as they do for pencil-and-paper assessments. These studies explored the interaction of assessment and instruction and call for rethinking the quantitative models of reliability and validity and work toward a more qualitative method of assessment that establishes reliability and validity. Koretz, Stecher, Klein & McCaffrey (1994), suggest the difficulty in establishing reliability and validity of an alternate assessment lies in the values of the state in defining student success. Koretz, et al. (1994) state a conflict that arises between the instructional and measurement goals that either embed assessment in the instructional process or rely on nonstandardized tasks affects the assessments reliability and validity.

Perhaps the lack of evidence of the technical quality of alternate assessment has resulted in the states hesitancy to include the test results in the accountability formulas. Since large-scale assessment is a relatively new advancement in special education, studies and recommendations to assess reliability and validity in large-scale assessment in special education are just now occurring (Tindal et al., 2003).

Recent research on alternate assessments has focused on examining teacher opinions about the portfolio (Kleinhert, Kennedy, & Kearns, 1999), correlating teacher opinion and student scores (Kampfer et al., 2001), and examining the impact of the IEP development (Kleinert & Kearns, 2001; Kleinhert, Kennedy & Kearns, 1997). Stiggins (2000) suggests that professional and subjective judgment is key to determining the quality of portfolio assessment. He proposes that a teacher's vision of what it means to be academically successful affects the results of the performance assessment. The teacher's understanding of the assessment and the student achievement is key to improved student achievement. Although this research has been somewhat beneficial when investigating the impact of portfolio assessments on instruction, it does not address the issue regarding the reliability and validity qualities of alternate assessment for use as part of the states accountability system.

Studies have investigated if the amount of time a teacher puts into preparing the alternate assessment, specifically a portfolio, affects the student's score. In 1999, Kampfer, et al. (as cited in Kleinert and Kearns, (2001) p. 221) surveyed all the special education teachers in Kentucky that had students participate in the alternate assessment. They investigated (a) the relationship between the amount of time spent working on the portfolios and the score; (b) which portfolio items required the most teacher effort; (c) to what extent teacher variables such as experience and training and instructional variables such as the amount of student involvement in the construction of the portfolio, portfolio elements embedded in instruction and the perceived benefit to the student predict the portfolio score; (d) and what aspects of the alternate assessment tend to be of most concern for teachers. The results indicate that instructional variables, teacher variables, and time spent on preparation accounted for 27.5% of the variance in scores. However, the instructional variables were the critical variables in predicting the variance in the scores accounting for 24.1% of the variance, not the amount of time spent on preparing

the portfolio or the teacher's experience. These results may indicate that teachers need more concrete strategies for embedding portfolio development into daily instruction and strategies to increase the student's involvement in the creation of the portfolio.

By the end of the 20<sup>th</sup> century, many states had developed alternate assessments as required by IDEA' 97 although many states were uncertain as to how they would incorporate the results of alternate assessments into accountability formulas (Quenemoen, et al., 2002). The passage of NCLB 2001 mandated states to develop a process so the results of the alternate assessments can be aggregated with those of the state general assessment. However, before those performance standards can be set, test developers need to assure that the alternate assessment is a valid measure based on the state concept. Messick (1989, as cited in Linn) outlined five types of evidence for determining validity: content, response process, internal structure, relations to other variables, and consequences. Miller and Linn (2000) discussed how Messick's framework applies to performance assessment and concerns with performance assessments such as the limited number of items assessed, the alignment of the assessment with the content standards, the confounding of constructs across content areas, and the influence of the scoring method on the construct and how these concerns should be addressed to collect validity evidence.

Content validity by itself is not considered sufficient evidence for assessing the validity of a test; however, the content evidence is important because the content of each portfolio varies based on the student's instructional program and the teacher's decision on what to include. Miller and Linn (2000) state that performance assessments present challenges with regard to content evidence because assessments are not always aligned with the content standards, and there are a limited number of tasks assessed which

narrows the definition of the construct. In a study of the Washington Alternate Assessment System, Johnson and Arnold (2004) found that the lack of construct validity in a portfolio assessment system may be due to teachers being unclear about how to select IEP skills that are connected to the state content standard areas. The student's curriculum may not be aligned with the state content standards because the state standards do not represent functional goals for the student. Each portfolio also contained a limited number of skills assessed to represent a content area. Therefore it raises the question if this limited number of skills assessed sufficiently represents the broad content area.

Educators and policymakers have questioned whether the alternate assessments are assessing the depth of the concept behind the content standard or simply the performance of a skill. Concerns focus on whether scores of an alternate assessment based on work place skills measure the same progress toward core academic content standards and the ability for these scores to be aggregated within a state's scores (Almond, et al., 2000). According to the Education Week *Quality Counts 2004*, thirty-two states assess students with severe disabilities on extended or expanded content standards, six use grade level content standards, two use functional skills not aligned to the content standards, and four use a combination of functional skills and content standards.

According to the American Educational Research Association (AERA) (2000), the response process of test takers can provide evidence concerning the connection between the construct and the performance of the test taker. In a portfolio assessment for students with severe disabilities, the teacher is responsible for assembling the data, therefore the student's score is based mainly on the teacher's presentation of the data and the

completion of the process. The teacher responds to the state directive to select skills based on IEP objectives, creating the environment and supports necessary for the student to practice and measure student progress, collects the data, and presents the data in a format designated by the state. The scores in a portfolio are open to interpretation including the appropriateness of the student's IEP goals, the effectiveness of the instructional program, the alignment of the core content standards to the curriculum, or the need to provide effective staff development for teachers on content based instruction (Johnson & Arnold, 2004). As a result it could be reasoned that the teacher's response process will have an enormous impact on the student's score.

The internal structure of a test can indicate the degree to which the relationships among test items and test components conform to the construct on which the proposed test score interpretations are based (AERA 2000, p.13). Therefore, the selection of a scoring method is key to understanding the inferences that can be made from the portfolio assessment (Miller & Linn, 2000). Alternate assessments are typically scored through use of a rubric. States use a similar approach to setting up a rubric in that there are common criteria used to measure quality and the level of success of the student portfolio. Thompson & Thurlow (2001) suggests the criteria should measure how well a student can perform target behaviors or skills; the level of independence a student demonstrates; how well the skills are generalized to different environments, with different people, or in different activities; and how appropriate the skills and activities are for a student (p.104). Skill generalization is an important assessment criteria since it provided evidence the student is working toward achieving at the same level of the non disabled peers and is meeting the core content standards.

interferent wild interferent of a second second product of the second second second second second second second

Kleinert and Kearns (2001) studied the importance of scoring, specifically the standards or performance criteria used, in the development of alternate assessments. More than 20% of the 44 national authorities surveyed stated that scoring criteria must not be scored in isolation but should be used concurrently to be meaningful (Kleinert & Kearns, 2001, p.215). Targeted skills should be evidenced across multiple settings and community settings, and age appropriate choices should occur in the context of activities with peers without disabilities are examples of cross scoring criteria. Alternate assessment scoring varies and reflects teacher generated checklists in which the teacher scores their own students and submit the scores to the state with no state oversight which may or may not be followed by a sample percentage of student work samples. Other states train teachers to score the student work, but not a teacher's own students or own district work, followed by a scoring of sample work. In other states, a regional panel reviews all portfolios with at least two independent scorers.

Scoring of portfolios is, according to Salvia and Ysseldyke (2001), "neither simple or straightforward" (p. 248) nor is score aggregation or score reliability issues resolved. Salvia and Ysseldyke suggest that attention to greater objectivity, less complexity, more scorer training, and greater comparability of portfolio contents are the keys to better practice (p. 257). Thompson & Thurlow (2001) suggested states need to be aware of the following challenges that may affect the validity and reliability of the assessments. Portfolio assessments may have lower reliability and comparability of results than other types of tests. They are more difficult to implement and present greater variability such as time spent on the development of the portfolio, training of teachers, and teacher support, and they are more difficult to score, more time consuming to score, and more subjective to score than standardized tests.

Creating portfolio assessments that are reliable and valid is costly and challenging when used to measure progress for students with disabilities. Measurement error is consistently greater than that associated with traditional tests (Koretz, et al., 1994). Portfolios typically integrate a variety of knowledge and skills to produce a product. The interdisciplinary characteristics of portfolio assessment tasks increases the probability that the severity of the disability will interfere with the validity of inferences made. For example, for students with processing to cognitive impairments it may be difficult to disentangle a reading skill from the vocabulary that is part of the standard, leading to the question of if the test assesses the determined outcome.

To assure greater validity and reliability, portfolios should contain pre-defined structure for the portfolios and require a number of entries and types of entries to assure all of the portfolios are scored using the same criteria (Olsen, Mead, & Payne, 2002; Quenemon, Rigley & Thurlow, 2002). Scoring criteria can be a direct measure of student achievement, a measure of system conditions necessary for student success, or a combination of the two. Unlike traditional large-scale tests with scoring criterion embedded in the test development process, portfolios tend to represent a continuum of scoring approaches as opposed to specific categories such as appropriateness for grade level and degree of difficulty. The scoring criteria should reflect the state's identified outcomes as evidenced through sample student work or in student response items on a test. Scoring criteria are often in the form of rubrics and include: variety in settings, student progress, ability to generalize, age appropriateness, alignment to the standards, amount of support, self determination, socialization opportunities, general education participation, and parent satisfaction (Thompson & Thurlow, 2001).

For the purposes of result comparison, Quenemoen, et. al. (2002) suggests six commonly used scoring criteria regardless of the assessment protocol used by the states. These common areas include: independence, generalization of skills, appropriateness of skills, performance, IEP linkage, and content standards linkage. Quenemoen, et.al (2002) compared and contrasted the assumptions and values embedded in scoring criteria used in five states for their alternate assumptions. Although these five states use different alternate assessment approaches, they share three basic quality indicators. Each state developed their alternate assessment through input from the stakeholders including educators, parents, researchers, and technical advisors reflecting professional and research based understanding of possible outcomes for students with significant disabilities. Each state continues to improve the reliability and validity of their assessment through documentation of the technical adequacy of the assessment. The states were also able to document a coherent alignment of the basic assumptions about teaching and learner outcomes. The methods used by the states reflect an understanding of best practices in alternate assessment leading to a level of confidence in the internal consistency and integrity by the researchers, thus providing a guide for other states' development of the alternate assessment.

# New Jersey Assessment and Accountability System

New Jersey has used statewide assessments as a means for student progress and program planning for many years, even before the federal legislation mandate. In 1975, the New Jersey legislature passed the Public School Education Act (PSEA) "to provide all children of New Jersey, regardless of socioeconomic status or geographic location, the educational opportunity which will prepare them to function politically, economically and socially in a democratic society" (N.J.D.O.E.). In 1976, PSEA was amended to establish uniform standards of minimum achievement in basic communication and computational skills. This amendment provided the legal basis for use of a test as a graduation requirement.

From 1978 through 1982, third, sixth and ninth graders participated in the Minimum Basic Skills (MBS) testing to assess reading and mathematics skills. Beginning in 1981, the MBS became one of the requirements for ninth grades to pass in order to graduate. In 1983, New Jersey adopted the Grade 9 High School Proficiency Test (HSPT9) to assess reading, writing and mathematics. The HSPT9 was administered as a graduation requirement during the 1985-86 school year.

In 1988, the New Jersey Legislature replaced the HSPT9 with the High School Proficiency Test (HSPT11) and added the Grade 8 Early Warning Test (EWT). The EWT was to serve as an assessment to assist in student placement and program planning, while the HSPT11 was a graduation requirement for all students.

The Core Curriculum Content Standards (CCCS) adopted by the New Jersey Department of Education in 1996 and revised in 2003-2004, identify what New Jersey students should know and be able to do by specific grades. The New Jersey's Assessment and Accountability system is multi-level series of test developed to determine how a student is progressing toward achieving the CCCS. New Jersey's statewide assessment system has been aligned with NCLB 2001. The statewide assessment system is referred to as the New Jersey Assessment of Knowledge and Skills (NJASK 3 and 4), the Grade Eight Proficiency Assessment (GEPA), the High School Proficiency Assessment (HSAP), the Special Review Assessment (SRA) and the New Jersey Alternate Proficiency Assessment ("NJAPA"). It mandates that all students in Grades 3, 4, 8, 11 take the relevant test for the designated grade level in the spring of the year and earn a proficient status on the H.S.P.A to receive a standard high school diploma. Students unable to participate in the NJASK, GEPA, HSAP, are required to participate in the NJAPA (N.J.D.O.E.)

The statewide assessment is designed to inform the New Jersey Department of Education and the public about how well schools and school districts are performing in teaching the CCCS. The test informs districts and parents about a student's proficiency in the tested subject area. The primary purpose of the New Jersey assessment program is to assess student achievement in reading, math, and science. A secondary purpose of the New Jersey Assessment is to compare the performance of students in New Jersey to student's performance across the country and within school districts in New Jersey by similar district factor grouping (DFG) status. Additional components of the plan include measuring student's performance, reporting school performance, and rewards for or sanctions for poor performing schools.

In accordance with NCLB 2001, New Jersey students with disabilities must participate in each subject area assessment of the NJASK, GEPA, and HSAP unless the student's

#### WARNING MA WORKS

IEP team determines that the student has not been instructed in any of the knowledge and skills tests and would not be able to do the types of items on the assessment (N.J.D.O.E.). The IEP team must decide what accommodations and/or modifications are necessary for the student to participate in the assessment and whether the student will take the general assessment or the NJAPA.

Students deemed unable to participate in the general assessment even with accommodations and/or modifications, participate in the NJAPA. The NJAPA is a portfolio assessment designed to measure progress for these students. Portfolio contents are individualized, linked to the CCCS, and may include a wide range of samples of student learning, including but not limited to actual student work, observations recorded by multiple persons on multiple occasions, test results, record reviews, or even video and audio records of student performance. The portfolio contents are scored according to predefined scoring criteria, usually through application of a scoring rubric to the varying samples of work.

In New Jersey, the majority of students with disabilities participate in general assessments as opposed to alternate assessments. In 2004, a total of 3,314 students took the APA with 3,039 students having valid Language Arts Literacy portfolios and 2,978 having valid Mathematics portfolios. New Jersey state results for the APA indicate that the percentage of students that scored at or above proficient on the Language Arts Literacy portfolios was: Grade 3, 91.7%; Grade 4, 92.9%; Grade 8, 91.3%; and Grade 11, 90.5%. Non-special needs districts scores and average of 93.5% at or above proficient on the Language Arts Literacy portfolios for Grades 3, 4, and 8 and 90.7% in Grade 11, while special needs district students scored an average of 87.6% in Grade 3, 91.4% in

Ly/org 2/ applicates to Etcologe

minute with dissidifiers. Education for all Finaling Art (2018) the net marketon can

Grade 4, 86.6% in Grade 8 and 89.6% in Grade 11. Mathematics APA portfolio results indicate 87.9% of students assessed at Grade 3 were at or above proficient with 4th Grade results at 88.5%, 8th Grades results at 89.4% and 11th Grade at 86.7%. Non-special needs districts scores and average of 90 % at or above proficient on the Mathematics portfolios for Grades 3, 4, and 8 and 85.7% in Grade 11 while special needs district students scored an average of 83.4% in Grade 3, 86% in Grade 4 and Grade 8 and 89.7% in Grade 11. Since this is the first year that scores were formally reported for public distribution caution should be used when concluding the success of the results of the testing program (N.J.D.O.E.).

### Effects of Inclusion in Testing

Over the past 30 years, major educational changes have occurred in the education of students with disabilities. Education for all Handicap Act (EHA) did not envision one placement for all students, but a full range or continuum of alternative placements for students with disabilities. P.L.94-142 defines least restrictive environment as the following:

...to the maximum extent appropriate, handicapped children, including those children in public and private institutions or other care facilities, are educated with children who are not handicapped, and that special classes, separate schooling, or other removal of handicapped children from the regular educational environment occurs only when the nature of the severity of the handicap is such that the education in regular classes with the use of supplementary aids and services cannot be achieved satisfactorily (P.L.: 94-142, section 1412[5][b])

Least restrictive environment is clearly defined in the law; however, interpretations of least restrictive environment vary considerably. This differs from the idea of inclusion

and mainstreaming. Least restrictive environment is the legal term whereas the term inclusion does not appear in the law. The term mainstreaming and integration, relating to student placement, also do not appear in federal laws or regulations. Even though the law has defined least restrictive environment, educators have created definitions for mainstreaming, integration and inclusion.

The educational initiative for least restrictive environment has raised the awareness of the need to educate students with disabilities within the general education setting. More students with disabilities are now being educated in general education settings with their peers without disabilities. Although state departments of education, and legislation have endorsed the educational movement toward least restrictive environment, recent court decisions reinforce the IDEA that "one size does not fit all." Schools vary in their placement of students with disabilities ranging from self-contained programs in separate schools to inclusion of student in general education classes. Ideally, students with disabilities should have full access to the general education curriculum and opportunities for interactions with typical peers. Theoretically, this should occur with the support of general educators and school administrators. However, without legislative mandate this was not always a reality (Kleinhert & Kearns, 2001).

Several important court cases have provided guidelines for including students. Daniel RR v. State Board of Education, 1989, Greer v. Rome City School District in Georgia, 1991, Sacramento City Unified School District v. Holland, 1992, and Oberti v. Board of Education of Borough of Clementon School District, 1993. The Oberti case provides the most relevant guidelines in New Jersey. The case allowed court to rule the following: School districts have an obligation to consider placing students with disabilities in regular education classes with supplementary aids and services before they explore other alternatives...Therefore the starting place for an inquiry into what the appropriate least restrictive environment for a student must be inclusion as a regular member of a regular education classroom; then the inquiry can then move to the matter of a more restrictive environment if it is found that that environment is not appropriate.

Simply assuring a student is placed in a least restrictive environment does not always guarantee the student has access to the general curriculum and therefore access to quality instruction and standardized assessment. The phrase access to the general curriculum was defined in IDEA '97 referring to the need for the education of students with disabilities to be based within the general education curriculum and standards.

To assure the inclusion of students in the general education curriculum, IDEA'97 stipulates that the planning and development of the IEP must reflect the student's participation in the general curriculum. The IEP must include the student's present levels of education performance and how the student's disability affects the student's participation and progress in the general education curriculum (614[d] [1] [A] [I] [aa]). IDEA 2004 further stipulates that for students with disabilities who take alternate assessments aligned to alternate achievement standards, a description of benchmarks or short-term objectives must be included (614[d] [1] [A] [I] [cc]). A statement of measurable annual goals, including academic and functional goals designed to meet the student's needs that result from the student's disability to enable the student to be involved and make progress in the general education curriculum (614[d] [1] [A] [I] [aa). The special education, related services, and supplementary aids must be designed to enable the student to be involved and make progress in the general education curriculum (614[d] [1] [A] [II] [aa).

nonacademic activities (614[d] [1] [A] [IV] [bb]). Through the present levels of educational performance, the goals and objectives and the provided services, a student with a disability is therefore assured involvement in the general education curriculum.

NCLB 2001 and the revisions to IDEA 2004 provide the legal mandates for the participation of students with disabilities in state and district assessments; however, the questions of how to accomplish this task and the benefits of this inclusion remains uncertain. Kleinhert and Kearns (2001) suggest at least three strategies for ensuring the inclusion of all students with disabilities in the general curriculum. The first strategy is to ensure that the student has the opportunity to learn their IEP objectives in the context of the general education classroom and to receive instruction in the context of the general education activities (p. 7). The student should have opportunity to participate in these activities in the home, the school and within the community.

Assuring a student participates in activities in the general education classroom does not assure a connection to the learning standards. Owen White's 1980 (as cited in Kleinert and Kearns, 2001, p.7) approach called for alignment of the IEP objectives with the learning standards through critical function. When assessing students with disabilities, the focus is on the function of the standard that will enhance the students' life. This approach allows the student to meet the learning standards by demonstrating acquisition of the skills, thus mastering the essential content of core learning for all students (Kleinert & Kearns, 2001, p.7).

The third strategy to enable students with disabilities to access the general curriculum is to address the standards through learning the academic content. Through an inclusive

curriculum design, students with disabilities would be a part of the general education classroom to learn at least some of the academic content (Kleinert & Kearns, 2001, p.8).

When the inclusion of students in state assessment systems was mandated, it was viewed as a process to assure students with disabilities would be included in local and state policy making decisions. The inclusion was viewed as a method to improve opinions about students with disabilities, improve access to the general education curriculum, and improve instruction in special education programs (Browder, Spooner, Algozzine, Ahlgrim-Delzell, & Karvonen, 2003). Research is beginning to show positive consequences of this inclusion although extensive studies on the effects of inclusion in statewide testing and the accountability process is limited due to the recent legislative requirements.

Data on the influence of alternate assessment results on policy decisions is difficult to determine at this time due to schools and states not having a clear picture of educational outcomes related to the inclusion of students in the assessment process. In a report on state reporting processes in 2001, Thompson and Thurlow (2001) reported that only half the states reported alternate assessment scores with the remaining states reporting that they were continuing to decide how to report the scores. The reporting of results has improved as reflected in Thompson and Thurlow's (2001) state report that indicates 96% of states reported student scores. At this point the influence of inclusion on policy and decision making is inconclusive until additional data and information is obtained.

Thompson and Thurlow's (2001) survey of state directors of special education found positive effects among states with an increase in the inclusion of these students in state assessments (90%). Questions surrounding the inclusion of students in the general

assessment or the alternate assessment have generated research in determining eligibility requirements for participation in alternate assessments. During the initial development of alternate assessments, 9 out of 12 states considered using cognitive level and adaptive behavior as precluding students from participation in the general assessment. Bechard's (2001) report on models for reporting results of assessments found that 20% of students with disabilities participated in the alternate assessment instead of the general assessment. Kleinert and Kearns (as cited in Browder, Spooner, Algozzine, Ahlgrim-Delzell, & Karvonen, 2003), reported that in Kentucky the alternate assessment rate of participation was much lower and used specifically for students with the most significant disabilities.

Although the inclusion rate has improved, questions remain as to whether participation has increased student expectations. Indirect evidence is reported in Kentucky by Kleinert and Kearns (1999, as cited in Browder, Spooner, et al., 2003). Scores on the alternate assessment have improved over the years with teachers reporting that the expectation for students to complete self-determination skills as part of the alternate assessment process has improved the expectation that these students are capable of self-assessing. The study reports an increase in the use of augmentative communication systems alluding to the expectation of the need for the students to communicate and express themselves.

An option for determining if alternate assessments have affected expectations for students is to review student IEPs. Thompson and Thurlow (2001), found in 41 states, only 8 states had no reference to the alternate assessment in the IEP and only 5 states addressed the state standards. Turner, et al. (2000) found no correlation between the

quality of the IEP and performance on the alternate assessment or a reflection of expectations on the IEP.

Relationship of the alternate assessment to the state's standards is another technique used to determine student expectations. Researchers found including academics rather than just life skills or functional skills in alternate assessments linked the state standards to the alternate assessment process (Kleinert & Kearns, 2001; Thomspson & Thurlow, 2001).

The premise that the use of alternate assessments will increase access to the general curriculum and assess all students on the state standards is not evident in the research. Browder, Spooner et al., (2003) found only 17% of states evaluate whether the skills included in the alternate assessment reflect state standards.

Data received from the alternate assessment to improve the quality of instruction is necessary to determine the benefits of the alternate assessment process. Thompson and Quenemeon et al. (2003) state the time invested in the alternate assessment may benefit both the student and the teacher. The information provided through the assessment results can be used to improve instruction thus improving student achievement.

There is currently insufficient research to state specific benefits of the alternate assessment on improved instruction. One study by Turner et. al (2000) correlated time investment to quality indicators and ultimately improved instruction and improved student outcomes. Others question if the amount of time spent on the preparation and collection of materials for the portfolio stresses teachers without necessarily improving instruction. DeStefano et. al. (2001) conducted a study the benefits of staff development focusing on the planning and participation in different assessments. The study showed improvement in the teachers' understanding of when to use the alternate assessment but not on how to use the results to improve instruction. Browder & Spooner, et al. (2003) described two methods to use the data for decisions regarding instruction. They stated that that unless teachers are specifically trained on data based decision making, the benefits to improved instructional process through use of the alternate assessment may not occur.

States are beginning to be able to document performance trends of students with disabilities. Forty of the fifty states (80%) report the ability to document performance trends (Thompson & Thurlow, 2001). The reported trends show overall improved state assessment performance as compared to the trend data from 1998 to 2001 that showed progress in reading and writing but minimal progress in math. Trends have been difficult to compare to past years of assessment progress due to changes in the assessment processs over the past several years.

Though the progress of students with disabilities in standards-based accountability has demonstrated improvement, states continue to face challenges in an effort to increase student achievement and administer assessments that provide documentation of this achievement. The major concern among 48% of the state directors of special education is that students with disabilities will not be able to achieve proficiency on state assessment. Twenty-six percent of the state directors believe the students cannot access or reach the states standards, while 28% believe that students with disabilities make school look less effective due to the low test scores. Other concerns include students being stressed by taking tests (40%), students with disabilities not graduating (20%), and to a lesser extent an increase in students identified for alternate assessments (12%) (Thompson & Thurlow, 2001).

### Summary

Students with disabilities are required by NCLB 2001 to participate and demonstrate improved performance in state assessments. The assessments are used to provide information on the educational progress of students and the extent to which students are achieving state standards. Since the start of alternate assessments, students eligible for these assessments ranged from students with severe and profound disabilities to some students with moderate and other disabilities. Gathering data on the performance of students through alternate assessments requires rethinking traditional assessment methods. Reporting on students with disabilities must be done in the same way and with the same frequency as for students without disabilities. IDEA 2004 reinforced these requirements, clarifying that states must report on both alternate assessments based on grade-level achievement standards and alternate achievement standards, and must also report on the number of students using accommodations to participate in the general assessment. States are now implementing accountability plans with consequences assigned to schools, administrators, teachers, and other educators. A review of the current research in large-scale assessment and accountability revealed that studies to date have not been definitive on the consequences of the assessment and accountability reform effort on students with disabilities with respect to the curriculum, instruction, parental and teacher awareness.

# Chapter III

Methodology

### Overview

This chapter describes the methodology of the study including the research design, participants, survey and focus group process, data collection procedures, and the analysis of the data. Qualitative and quantitative research were chosen to analyze the insights, attitudes and experiences of directors, supervisors, or coordinators of special education programs in public school districts in New Jersey toward the consequences of large-scale assessment results for students with disabilities. The results provide an understanding of the initiatives, trends, accomplishments, and emerging issues involving the New Jersey alternate assessment process. A second purpose was to determine if the accountability requirement has had an effect on curriculum, instruction, and special education placement.

A review of the current research in large-scale assessment and accountability reveals that studies to date have not been definitive on the consequences of the assessment and accountability reform effort on students with disabilities with respect to the curriculum, instruction, parental and teacher awareness. This study explored directors of special services' perceptions regarding the consequences of high-stakes assessment for students with disabilities, with specific reference to the New Jersey Alternate Proficiency Assessment (NJAPA) and the impact of accountability on special education services.

The NJAPA assesses the achievement levels of New Jersey's students with the most severe and cognitive disabilities and is the basis for New Jersey's accountability system relative to these students. The primary intent of the New Jersey assessment system is to improve student achievement levels and increase accountability for school districts in assuring adequate yearly progress and quality instruction. The outcomes of the assessment policy are considered indicators of how New Jersey's assessment program is achieving its intended purpose, as well as how it is functioning beyond its intended purpose of student achievement. Evidence of improved student achievement is an indicator used to determine a school's ranking and the School Report Card and as a performance based assessment for funding.

Although test scores are indicators of student achievement, they cannot alone measure the impact of the New Jersey assessment process. The literature indicates large-scale assessments and accountability systems impact a variety of outcomes including student learning, instruction, curriculum development, student, teacher and parental awareness, and professional development for instructional and administrative staff. Most research reflects national trends or is relevant to specific states. Therefore, research conducted within the central and northern New Jersey region has been conducted to identify consequences that are unique to this area as well as trends that are reflective of other state and national movements.

# Methodology

The descriptive study employed two methodologies: administering a survey (quantitative design) and conducting a focus group session (qualitative design), to explore directors of special services' perceptions regarding the consequences of high-stakes

assessment for students with disabilities, with specific reference to the NJAPA and the impact of accountability on special education services.

### Research Design

Evidence to evaluate large-scale assessment systems has been identified by the American Education Research Association (2000), The National Center on Educational Outcomes (Thompson & Thurlow, 2001) as well as individual research (Chudowsky & Behuniak, 1998; Lane, Parke, & Stone, 1998; Meherns, 1998). Lane et al. (1998) suggests the first step in examining the consequences of an assessment system is to identify the intended effects on such variables as the curriculum, the instructional contents and strategies, and content and format of classroom assignments, and so on. Examining the intended positive effects will likely lead to the identification of negative effects and unintended consequences.

According to Fink (1995, p. 1), the survey is a direct research method used for collecting information to describe, compare or explain knowledge, attitudes, and behavior. Rea and Parker (1997) state that "there is no better method for determining, with a known level of accuracy, information about large populations" (p.3). A survey should be designed to meet the objectives of the research obtain the most complete and accurate information possible; and to be completed within the limits of available time and resources (Sheatsley, as cited in Rossi, Wright, Anderson, 1983). Based on Fraenkel's (1996) and Linsky, (1975) submission that a survey instrument should be brief enough to encourage the subjects to complete and return it, the researcher attempted to word the survey items clearly and briefly so that the participants could easily understand the question. Assuring the results obtained from the sample population can be generalized to the "target population" is extremely important when defining the population of the study (Anderson, 1999, p.196). The researcher must also select the sample population that has information the researcher is seeking and is willing to respond to the survey (Fraenkel, 1996).

The Perceptions of Accountability in Special Education (PASE) is a Likert type scale consisting of 15 questions. Questions focused on student instruction, testing accommodations, alternate assessments, accountability, and current and emerging curricular practices and issues. Each respondent was asked the same set of questions to ensure reliability.

The choice of the demographics items was included in the survey since peoples' perceptions are affected by demographic factors such as their sex, age, level of education, socioeconomic factors, ethnicity, and contact (Sheatsley, as cited in Rossi, Wright, Anderson, 1983, p. 204). Demographic questions in this study's survey include personal information, such as years of experience as a director and gender. District demographic questions deal with the district size, total number of special education students as of the official 2004 count in New Jersey and the District Factor Group.

The District Factor Group is an indicator of the socioeconomic status of a district's population. The measure was developed in 1974, based on the 1970 census. Revisions occurred in 1984 and 1992 as a result of the changes in the 1980 and 1990 census. The variables considered by New Jersey include: percent of adult residents who failed to complete high school, percent of adults who attend college, occupational status of household members, population density, income, unemployment and poverty. After

many mode in objectives when the edgebrackies between the convergent the purpose of the study is evolved (forwise, 1953). Acception of the scarple population can also denote  $_{62}$ 

applying a formula known as the principal components analysis, a single socioeconomic measure is obtained for each district. The districts are broken down into eight categories, A, B, CD, DE, FG, GH, I, J, with A the lowest socioeconomic district and J the highest (N.J.D.O.E.). For the purpose of this study, the groups were ranked low (A, B), middle (CD, DE, FG, and GH), and (I, J) high.

Issues considered in the survey design included the clarity of the directions and questions, the length of the survey and the order of the items. According to Babbie (1999), the order impacts the less educated more than the more educated. Since the population surveyed reflects a highly educated population, Babbie's suggestion is not paramount to the design of the survey. The answer to a question is valuable to the extent that it can show a predictable relationship to facts or subjective areas of interest. A survey meets its objectives when the relationship between the answers and the purpose of the study is evident (Fowler, 1993). Selection of the sample population can also affect the results of a survey even if it appears the selected questionnaire is the most suitable for the research purpose. In a survey, it is assumed that the respondents have a general knowledge of the subject. However, Sheatsley's (as cited in Rossi et al., 1983) suggestions that if the population does not have the information, they cannot recall it, cannot predict it, or are ignorant in the subject matter, the end result may not be applicable to the research purpose (p. 196). Since this research directly implicates a subset of a larger population, the concerns related to a lack of knowledge of the subject area has been controlled (Fink, 1995, p.27, Fraenkel, 1996). The sample group has a clearly defined eligibility in that the participant must be a director, supervisor, or coordinator of a special education program in a public school district in New Jersey with

an administrative and functional knowledge of special education services. The willingness or readiness of the respondent to reply may also impact the findings. Careful introduction of the survey, proper survey auspices, and a well-planned line of questioning that does not depend on a few blunt questions add to the validity of the questionnaire (Sheatsley, as cited in Rossi et al., 1983).

Fink (1995) describes validity of an instrument as being accurate, while reliability of an instrument is described as being consistent. The relationship of the intent of the questions to the study's purpose, and the consistency of the responses across the reviewed surveys were followed to assure a relationship to the purpose of the study.

# Focus Group

In addition to the survey, the researcher conducted a focus group to add to the richness of the data. Because the subject of accountability and assessment are both sensitive and complex, the choice of a survey and focus group was chosen. Survey researchers frequently are concerned that the use of one research method such as a survey limits the study results and a less acceptable study. Campbell as cited in Rossi, et al., (1983) advocates for the use of multiple methods in the measurement of social variable. The use of triangulation or the method of combining methodologies is in the words of Patton, (1990) "Ideal" (p.187). Patton states "studies that use only one method are more vulnerable to errors linked to the particular method (e.g., loaded interview questions, biased or untrue responses) than studies that use multiple methods in which different types of data provide cross-data validity checks" (p. 188).

Chudowsky and Behuniak (1998) recommended focus group methodology in addition to survey methods for collecting evidence regarding consequences of testing systems in that focus groups (a) allow for unanticipated responses, (b) allow for the researcher to gain an understanding of the nature and strength of the emotional reactions to participants, (c) provide illustrative points and added interest to the reader, and (d) produce insights by the nature of the interaction of the group members.

Through focus groups, the researcher derives understanding based on the discussions, as opposed to the testing of a theory or hypothesis. Focus groups may be defined as group interviews that bring together a small number of participants from a well-defined target population to discuss a set of preselected topics under the guidance of a moderator (Krueger, 2000). Focus groups typically elicit a wide range and depth of responses and are often used to evaluate ongoing programs (Patton, 1990). Information obtained from the focus group can be maximized by proper selection of the focus group participants, planning of the group interview time, clarification of any issues, and systematic analysis of the information obtained during the discussion.

The questions for the focus group were determined and sequenced based on an analysis of the literature, focus group questions from previous doctoral dissertations, and based on open-ended survey questions from the National Council for Educational Outcome research. The question guide was arranged in a logical manner to facilitate group discussion with each type of question designed to provide feedback relative to the research questions. Questions were open-ended and non-specific so that the participant can interpret the questions based on his or her own experiences. Opening questions and introductory questions were designed to establish trust with the group members and
provide an understanding as to why they were an essential part of the group. Ending and summary questions provided a check to the interviewer the participants perceived responses. These questions also provided an opportunity to identify the most important aspect of their beliefs. Probes for clarification of vague and ambiguous responses, probes for elaboration on potentially important points, and silent probes were utilized (Murphy 1980).

The researcher's professional relationship was considered when developing the procedure for conducting the focus group. Professional contact with the directors of special services occurs occasionally during state and county special education meetings and during the placement of a student at the researcher's school. Professional contact with the directors of special services will not compromise the information obtained from the data. To reduce bias, participants were given a code that was referenced when recording of the observations. Participants were selected who represent diverse district factor groupings will reduce participant conformity. Providing a post survey so participants had the opportunity to reflect on their individual experiences in the event they were unable to respond during the discussion or to relay concerns they were unable to respond to during the group will reduce bias.

#### Sub jects

#### PASE Survey Participants

According to the New Jersey Department of Education, there are 118 districts in Essex, Middlesex, Morris, Somerset, Sussex, Union, and Warren counties in central and

65

Northern New Jersey. Districts in central and northern New Jersey were selected based on the district's 2004-2005 student enrollment, the District Factor Group (DFG), the district's listing as a K-8 or K-12 school district, and the district's participation in the New Jersey Assessment of Skills and Knowledge for Grades three-four (NJASK 3-4), the Grade Eight Proficiency Assessment (GEPA), and the Alternate Proficiency Assessment (APA).

Of the original 118 districts, four participated in the pilot study and 14 districts of the 118 did not meet the determiner requirements and were eliminated as possible participants. Surveys were sent to 100 directors of special services representing each department of special services in the designated districts. The information about the districts was obtained from the New Jersey Department of Education website. If available, the surveys were directed to the specific director of special services. The distribution of the groupings in this study are as follows: 13 districts in DFG Group AB; 59 districts in DFG Group CD, DE, FG, GH; and 46 districts in DFG

Group I, J.

Although the position of the director of special services for each district is an administrative one requiring at least a supervisory certificate, there are many titles for this position. The title is indicative of the director's background. A background in special education as a teacher, supervisor, child study team member, learning disabilities/teacher consultant (LDTC), or a principal with special education background comprised the participants' backgrounds. Director of special education, director of special services, director of pupil personnel services, director of personnel services, supervisor of special education, coordinator of special services, coordinator of child study team, and coordinator of special education are the common titles for this position in New Jersey. Even though the title differs among districts, the responsibility of operations for the district's special education program was the same.

# Focus Group Participants

Participants of the focus group were directors of special services who have direct contact with the members of IEP team, including parents, case managers, general education and special education teachers. The directors are also responsible for monitoring curriculum and instructional implementation and reporting and analyzing reports and scores. The focus group was comprised of participants from K-8 and K-12 districts. The group was homogeneous in nature with respect to certification, but heterogeneous with respect to years of experience in education and DFG (Krueger, 2000).

To standardize the selection process, participants were chosen from a list of directors of special services from central and northern New Jersey. Criteria for participation included the following:

- 1. a valid New Jersey certificate as a principal, supervisor, administrator;
- 2. direct responsibility for supervision of IEP teams;
- 3. direct responsibility for reporting and analyzing assessments scores and data;
- 4. direct responsibility for program changes and implementation; and
- 3 or more years experience in a supervisory position preferable special education

## Data Collection

#### PASE Survey

The PASE survey was sent with a 2-week return time. A follow-up mailing was sent out with a 1-week deadline. A cover letter, self-addressed stamped envelope, and survey was included in the mailing. A separate self-addressed post-card was included with the initial survey for the participant to return if they want a summary of the results of the survey. The follow-up mailing included a different cover letter, no postcard, another survey, and self-addressed envelope.

Respondent's names were not requested on the surveys. In order to maintain anonymity of participants, a code number was written on the back of each survey. Each survey was coded numerically for the sole purpose of identifying directors who did not respond so that a second survey could be sent. As completed surveys were received, respondents were deleted from the mailing list. Surveys were returned to a P.O. Box located at the local post office of the researcher's place of work. A master list and all responses were maintained in a locked file located in the researcher's office to protect the confidentiality of the survey respondents.

Response bias and nonresponses were minimized through the selection of the sample population (Fraenkel, 1996). Consideration was given to inducing respondents to complete the questionnaire without intervention of the researcher; however, an incentive was not included in the survey mailing.

# Focus Group

The researcher contacted potential focus group participants by mail and e-mail 10 to 14 days before the scheduled focus group. A confirmation letter or e-mail was sent 1 week prior to the focus group to confirm participation.

The focus group was recorded through written notes (as explained in the Letter of Solicitation and Informed Consent) and transcribed by the researcher. A summary of the collected data appears in Table 18 in chapter 14. Participants had the opportunity to verify the summary statement posed at the end of the session and through a summary of the discussion mailed to the participants.

#### Instrument Design

#### PASE Survey Design

A number of strategies can be used to improve validity and reduce bias with survey and focus group methodology. Systematic analysis of data is essential, from sequencing questions, capturing and coding of data, to participant verification of data.

The original survey PASE of 20 items and seven demographic questions was sent to a jury of experts. Included was the cover letter explaining the purpose of the research and the role of the jury of experts. The jury of experts was composed of four directors of special services who have been in the field of special education for 10 plus years and hold a doctoral level degree.

Each participant was asked to rate the items in the following manner:

1. Questions that you believe will reveal a director's perception toward the

nale of the pilot group. Thus of the directors were chocatorized of the jury of experim. This corponess and ted to divisiopling the final <u>status</u> consety. The following genericos

alternate assessment process and accountability were to be given a + (plus).

2. Questions that you believe will not reveal a director's perception toward the alternate assessment process and accountability were to be given a - (minus).

3. Questions about which you are unsure were given a 0.

The results of this rating system were complied with a + (plus) receiving 2 points, a 0 receiving 1 point, and a - (minus) receiving 0 points. Survey questions were chosen and modified as a result of the jury of experts comments.

In addition, two open-ended questions were asked. The first asked for suggestions for the survey and the second for any other comments.

From the original survey, 15 items were chosen and modified as result of the jury of experts' comments. A pilot survey was sent to 10 directors of special services in selected districts. This mailing included a cover letter explaining the purpose of the study and the role of the pilot group. Two of the directors were also members of the jury of experts. The responses assisted in developing the final PASE survey. The following questions were asked to the pilot group:

- 1. How long did it take you to complete the survey?
- 2. Were the directions clear?
- 3. Were there any survey items that were unclear?
- 4. Should anything else be included in the survey?
- 5. Was the layout of the survey clear?
- 6. Were there any other comments?

Hented on the resourch quantities, there not hypotheses were developed to gride the

# Focus Group Design

A third party reviewer reviewed questions for appropriateness prior to the focus group meeting. Introductory questions were open-ended and designed to allow the participants to reflect on their own experiences and understanding of the problem that was being discussed. These "key questions" (Krueger, 2000) were open-ended and required the researcher to utilize probing to elicit more in-depth information.

#### Data Analysis

# Analysis of the PASE Survey

Quantitative analysis procedures included basic descriptive statistics: measures of central tendency (mean and median) and dispersion (standard deviation and range).

Based on the research questions, three null hypotheses were developed to guide the quantitative portion of the research study.

1. There is no significant difference in the director of special services perception of accountability among the district size groups.

2. There is no significant difference in the directors of special services perception of accountability among district socio-economic status groups (DFG).

3. There is no significant difference in the directors of special services perception of accountability among the experience level groups.

Each hypothesis was tested using analysis of variance (ANOVA).

# Analysis of the Focus Groups

Qualitative analysis included the coding of directors of special services comments according to themes and patterns. The researcher's professional judgment was used in coding based on an analysis of words used in the focus groups (word repetitions, keywords in context, and professional terms).

# Summary

With approval from the Seton Hall University Institutional Review Board granted on July 26, 2005 the researcher used two instruments for determining the insights, attitudes and experiences of directors, supervisors, or coordinators of special education programs in public school districts in New Jersey toward the consequences of large-scale assessment results for students with disabilities. A second purpose was to determine if the accountability requirement has had en effect on curriculum, instruction, and special education placement. The first instrument was the PASE survey. The second instrument was a focus group session. It is believed that the results from the sample survey participants and the focus groups participants will result in conclusions that may be generalized to the larger population of directors of special services. the state of the second s

73

#### Chapter IV

# **Research Findings**

#### **Overview**

The purpose of the study is to explore the directors of special services perceptions of the consequences of high-stakes assessment and its impact on the increased demand for accountability on special education services in 100 public school districts in central and northern New Jersey. This chapter summarizes the results of the data analysis for the research questions posed in this study:

1. What is the perception of the directors of special services toward changes in student achievement and instruction of special education students since the inclusion of students with disabilities in the New Jersey state assessment system?

2. What is the perception of accountability in special education of directors of special services in low socioeconomic districts (DFG Groups) as compared to the perception of accountability in special education of directors of special services in high socioeconomic districts (DFG Groups)?

3. What is the perception of accountability in special education of directors of special services in high-enrollment districts as compared to the perception of accountability in special education of directors of special services in low-enrollment districts?

4. What is the perception of accountability in special education of directors of special services with less experience as compared to the perception of accountability in special education of directors of special services with more experience?

5. What is the of perception of the directors of special services toward changes in use of testing accommodations since the inclusion of students with disabilities in the New Jersey state assessment system?

6. What is the diversity of perceptions of the directors of special services toward changes in professional development since the inclusion of students with disabilities in the New Jersey state assessment system?

7. What is the diversity of perceptions of the directors of special services toward the effectiveness of the Alternate Proficiency Assessment since the implementation of New Jersey's state alternative assessment system for students with severe disabilities?

8. What is the diversity of perceptions of the directors of special services toward the effect of including students with disabilities in the New Jersey statewide assessment program?

Data for this study is presented by methodology. This descriptive study employed two research methodologies to answer the research questions: administering a survey (quantitative design) and conducting a focus group session (qualitative design).

The quantitative findings of the research are presented first. The return rate and comments from the Perceptions of Accountability in Special Education (PASE) surveys are discussed. Overall descriptions of the results are provided.

The qualitative findings were examined according to whether directors of special services viewed accountability as positive, neutral or negative. Table 18 encapsulates this information based on the responses to the questions posed during the focus group sessions.

#### Quantitative Research Results

According to the New Jersey Department of Education, there are 118 districts in central and northern New Jersey. Of the original 118 districts considered for participation in the study, four participated in the pilot study, 14 districts did not meet the determiner requirements (see previous definition) and were eliminated as possible participants. Therefore, 100 PASE surveys were sent (Attachment A). Of this number, 68 were returned in time for inclusion in this study (return rate of 68%). One survey was not included in the data analysis due to incomplete responses from the participant.

Research Question # 1 Frequencies of Perceptions of Directors of Special Services Toward Changes in Student Achievement and Instruction of Special Education Students Since the Implementation of the New Jersey's State Assessment System

The average response to the PASE survey ranged from an average low score of 1.86 to a high score of 4.43. The breakdown of frequencies is shown in Table 1, clustered as follows: disagree with values 1-2 (21); neutral with a value of 3 (37); and agree with values of 4-5(9). Responses (55.3%) of the fell in the neutral response area.

# Table 1

Summary of Frequencies for Changes in Student Achievement and Instruction

	Frequency	Percentage
Disagree	21	31
Neutral	37	55
Agree	9	13
This percentag	ge had been roun	ded and
therefore do n	ot total 100%	

The average score for all the PASE survey scores was 3.29, suggesting that there has been no noticeable change in student achievement and instruction for student with disabilities since their inclusion in the New Jersey state assessment system. However, with disabilities some availably railed progress toward the NICCCS when taking the general education accelerate or the ATA.

item analysis indicates that survey questions 4 and 15 pertaining to research question 1 indicates agreement, average score for all the PASE survey scores was 4, indicates that the directors' perception of improvement in the special education teachers' knowledge of the general education curriculum since the inclusion of students with disabilities in the New Jersey Assessment system. Results also indicate that directors agreed, average score for all the PASE survey scores was 4, that other alternative assessment methods than the current system that would accurately assess student progress for students with disabilities. Survey questions 2, 9, and 11 indicate a neutral response; average score for all the PASE survey scores was 3, of how directors perceive an improvement in general education teachers' ability to modify the curriculum when students with disabilities are included in their classes. Directors' answers indicated a neutral response that students with disabilities scores accurately reflect progress toward the NJCCCS when taking the general education assessment or the APA.

# Research Question #2 Director's Perception Toward Accountability of Students with Disabilities with Respect to District Factor Group

This question sought to determine if a difference in perceptions existed among directors toward accountability for student with disabilities with respect to district factor group (DFG). For this study, DFG groups were identified as Group AB, Group CD,DE,FG,GH,H, and Group IJ.

The breakdown of responses for the DFG group is provided in Table 2

		Solar Conference Internet Exclusion Lance Constant Taxana Antonio		77
Table 2	10 J.11/10 34552 J.044 23 353/03 2371123 J.060	0 1.2000 3.2208 9 1.2000 1.2709	2.48	1.26 0.38
Summary of Rec	eived Surveys with Respect to	DFG	2,47	
Summury of Kec	DFGGroup	Total Surveys	2,47	8.20 8.20
Summary of Kec	DFGGroup	Total Surveys Received	2.47 2.60 mill Junu 1	8:20 8:20 doi: 1155
Summary of Kec	DFGGroup AB CDDEFGGHH	Total Surveys Received 9 30	2.47 2.03 mill iturn, 1	820 820

The PASE Score and DFG Group's number of cases, mean, standard deviation, standard error, 95 % confidence interval and minimum and maximum, for each dependent variable and independent variables is described in Table 3. The average score for the DFG group responses was 3.29 for the PASE score.

# Table 3

# Descriptive Statistics for Accountability by DFG Group PASE SCORE

			95% Confidence Interval for Mean						
	N	М	SD	SE	Lower Bound	Upper Bound	Min	Max	
AB	13	3.1713	.34682	.09619	2.9617	3.3809	2.80	3.86	
CDDEFGG HH	29	3.3783	.37182	.06904	3.2368	3.5197	2.40	3.93	
IJ	25	3.2659	.46914	.09383	3.0722	3.4595	2.47	4.20	
Total	67	3.2962	.40858	.04992	3.1965	3.3958	2.40	4.20	

Hypothesis for question 2. The hypothesis for this question is stated in null form. Ho: The average perception of accountability score is the same for all three district socioeconomic status groups. The formula for the hypothesis (Ho: $\mu$ 1= $\mu$ 2= $\mu$ 3) states that there is no significant difference in the average perception of accountability between schools from different socioeconomic status groups as defined by the district's DFG Grouping. If the null hypothesis is rejected and H1 is accepted, this would mean that there is a significant different schools from difference in a director's perception of accountability between schools from different

	Som of		Pitton System	7	
Rational Controls	.721		201	1.272	78
socio-economic stat	us groups as	defined by the	district's DFG (	Grouping. The	.05 alpha

level of significance was used for determining rejection of Ho and acceptance of H1.

To determine the outcome for this hypothesis, an analysis of variance (ANOVA) was performed on the data and an F value of 1.272 was obtained. Results are reported in Tables 4 and 5.

# Table 4

Test of Homogeneity of Variances for Accountability by DFG Group

PASE SCORE			
Levene Statistic	df1	df2	Sig.
1.437	2	64	.245

#### Table 5

# ANOVAf or Accountability by DFG Group

PASE SCORE

	Sum of Squares	df	Mean Square	F	Sig.	
Between Groups	.421	2	.211	1.272	.287	
Within Groups Total	10.597 11.018	64 66	.166			

Levene's Test of Homogeneity with a significance level of .05 or better was used to test if there were equal variances across samples. Levene's Test of Homogeneity is not significant (.245) indicating the assumption of homogeneity for the variances has not been violated. The one-way ANOVA results for the main effect of DFG Group on accountability as indicated by the average PASE score is not significant with an F value of 1.272 (df=2,64, p=.287). The F value of 1.272 indicates the research fails to reject the null hypothesis and accepts that there is no significant difference in the directors of special services perception of accountability with respect to schools from different DFG

	LOG Servers
Sumo Con sel	Received
500-800	
0001-1000	
1001-3000	23
and entry and	10

Groups. This implies that DFG Grouping does not have a significant effect on perception

of accountability as indicated by the average PASE score.

Research Question #3 Director's Perception Toward Accountability of Students with Disabilities with District Enrollment as per the 2004 ASSA Report

This question sought to determine if a difference in perceptions existed among directors toward accountability for student with disabilities with respect to district enrollment as per the 2004 ASSA Report. For this study the district enrollment groups were defined as: 500-800, 801-1000, 1001-3000, 3001-5000, and more than 500.

The breakdown of responses for the district enrollment groups is provided in Table 6.

Table 6

Summary of Received Surveys with Respect to District Enrollment Groups

District	Total Surveys	
Enrollment	Received	
500-800	8	
801-1000	7	
1001-3000	23	
3001-5000	19	
5001 and more	10	

The PASE score and District Enrollment Group's number of cases, mean, standard deviation, standard error, 95% confidence interval, and minimum and maximum, for each dependent variable and independent variables is described in Table 7. The average PASE score for the District Enrollment group responses was 3.26.

# Table 7

PASE SCOR	Е					e uni zacol		
					95% Confider for M	nce Interval ean		
	N	М	SD	SE	Lower Bound	Upper Bound	Min	Max
500-800	8	3.2500	.52705	.18634	2.8094	3.6906	2.53	3.93
801-1000	7	3.5048	.34610	.13082	3.1847	3.8249	2.87	3.93
1001- 3000	23	3.2812	.42733	.08910	3.0964	3.4659	2.40	4.20
3001- 5000	19	3.3228	.42703	.09797	3.1170	3.5286	2.47	4.00
5001- 9000	10	2.9600	.29178	.09227	2.7513	3.1687	2.60	3.47
Total	67	3.2647	.42876	.05238	3.1601	3.3693	2.40	4.20

Descriptive Statistics for Accountability by District Enrollment Group

*Hypothesis for question 3.* The hypothesis for this question is stated in null form. H<sub>0</sub>: The average perception of accountability score is the same for all five district size groups. The formula for the hypothesis (Ho: $\mu$ 1= $\mu$ 2= $\mu$ 3= $\mu$ 4= $\mu$ 5) states that there is no significant difference in the average perception of accountability between schools of different enrollment sizes. If the null hypothesis is rejected and H1 is accepted, this would mean that there is a significant difference in a director's perception of accountability between schools from different enrollment groups as defined by 2004 ASSA report. The .05 alpha level of significance was used for determining rejection of Ho and acceptance of H1.

To determine the outcome for this hypothesis, an analysis of variance (ANOVA) was performed on the data and an F value of 2.028 was obtained. Results are reported in Tables 8 and 9. Litrene's Test of Hearogeneity with a significance level of .15 as Mithr was used to

that if there were putted variances average numples, however's Tost of Harrong study is will 81

#### Table 8

Test of Homogeneity of Variances for District Enrollment Group

lfi	df2	Sig.
		.401
1	if1	ifi df2

## Table 9

# ANOVA for District Enrollment Group

PASE SCORE

	Sum of Squares	df	Mean Square	F	Sig.
Between Groups	1.404	4	.351	2.028	.101
Within Groups	10.729	62	.173		
Total	12.133	66			

Levene's Test of Homogeneity with a significance level of .05 or better was used to test if there were equal variances across samples. Levene's Test of Homogeneity is not significant (.401) indicating the assumption of homogeneity for the variances has not been violated. The one-way ANOVA results for the main effect of district size on accountability as indicated by the average PASE score is not significant with an F value of 2.028 (df=4, 62, p=.101). The F value of 2.028 indicates the research fails to reject the null hypothesis and accepts that there is no significant difference in a director of special services perception of accountability with respect to schools of different enrollment sizes. This implies that district size does not have a significant effect on the perception of accountability as indicated by the average PASE score.

	hister/add
L-D Webby	13
A-S tealers	27
10 or there weat	28

Research Question #4 Director's Perception Toward Accountability of Students with Disabilities with Respect Years of Experience in the Position of Director of Special Services

This question sought to determine if a difference in perceptions existed among directors toward accountability for student with disabilities with respect to years of experience in the position of director of special services. For this study the groups were defined as less than 1 year, 1-3 years, 4-9 years, and 10 or more years.

The breakdown of responses for the years for the years of experience in the position of director of special services group is provided in Table 10.

Table 10

Summary of Received Surveys with Respect to Years of Experience

Total Surveys
Received
12
27
28

The PASE score and Years of Experience group's number of cases, mean, standard deviation, standard error, 95% confidence interval, and minimum and maximum, for each dependent variable and independent variables is described in Table 11. The average PASE score for the Years of Experience group responses was 3.26.

# Table 11

# Descriptive Statistics for Accountability by Years of Experiences

PASE SCORE

N M SD		SE	95% Confidence Interval for Mean		Min	Max		
					Lower Bound	Upper Bound		
1-3 YEARS	12	3.2500	.41500	.11980	2.9863	3.5137	2.47	3.80
4-9YEARS	27	3.2790	.45179	.08695	3.1003	3.4577	2.40	4.20
10ORMORE YEARS	28	3.3595	.33913	.06409	3.2280	3.4910	2.80	3.93
Total	67	3.3075	.39780	.04860	3.2104	3.4045	2.40	4.20

*Hypothesis for question 4.* The hypothesis for this question is stated in null form. H<sub>0</sub>: The average perception of accountability score is the same for all four directors of special services experience level groups. The formula for the hypothesis (Ho: $\mu$ 1= $\mu$ 2= $\mu$ 3) states that there is no significant difference in the average perception of accountability between schools of different enrollment sizes. If the null hypothesis is rejected and H1 is accepted, this would mean that there is a significant difference in a director's perception of accountability between director's years of experience in the position of director of special services. The .05 alpha level of significance was used for determining rejection of H0 and acceptance of H1.

To determine the outcome for this hypothesis, an analysis of variance (ANOVA) was performed on the data and an F value of .426 was obtained. Results are reported in Tables 12 and 13.

84 Table 12 Test of Homogeneity of Variances for Years of Experience PASE SCORE Levene Statistic dfl df2 Sig. .264 1.359 64 2 Table 13 ANOVA of Years of Experience PASE SCORE Sum of df Sig. Mean Square F Squares Between Groups .426 .655 .137 2 .069 Within Groups 10.307 64 .161

10.444

66

Total

# Levene's Test of Homogeneity of variances with a significance level of .05 or better was used to test if there were equal variances across samples. Levene's Test is not significant (.264) indicating the assumption of homogeneity for the variances has not been violated. The one-way ANOVA results for the main effect of years of experience in the position of director on accountability as indicated by the average PASE score is not significant with an F value of .426 (df=2, 64, p=.655). The F value of .426 indicates the research fails to reject the null hypothesis and accepts that there is no significant difference in a director of special services perception of accountability with respect to years of experience in the position of director. This implies that years of experience in the position of director does not have a significant effect on the perception of accountability as indicated by the average PASE score.

Research Question# 5-Frequencies of Perceptions of the Directors of Special Services Toward Changes in Use of Testing Accommodations Since the Implementation of New Jersey's State Assessment System

The average response to the PASE survey ranged from an average low score of 2 to a high score of 5. The breakdown of frequencies is shown in Table 14, clustered as follows: disagree with value 2 (10), neutral with a value of 3 (11), and agree with values of 4-5(46). Responses (68.6%) fell in the agree response area.

Table 14

Summary of Frequencies with Respect to Changes in Use of Testing Accommodations

	Frequency	Percentage		
Disagree	10	15%		
Neutral	11	16%		
Agree	46	69%		

The average score for all the PASE survey scores was 3.67 suggesting that there has been no noticeable change in curricular accommodations with the adaptations continuing to be used for instruction and testing.

Research Question # 6- Frequencies of Perceptions of Directors of Special Services Toward Changes in Professional Development Since the Implementation of the New Jersey's State Assessment System

The average response to the PASE survey ranged from an average low score of 2 to a

high score of 5. The breakdown of frequencies is shown in Table 15, clustered as

follows: disagree with value 2 (21), neutral with a value of 3 (14), and agree with values

of 4-5(32). Responses (47.8%) fell in the agree response area.

# Table 15

# Summary of Frequencies with Respect to Professional Development

	Frequency	Percentage
Disagree	21	31
Neutral	14	21
Agree	32	48

The average score for all the PASE survey scores was 3.22 suggesting that there has been some noticeable change in professional development since the inclusion of student with disabilities in the New Jersey state assessment system but the effect of the

professional development is unknown at this time.

Research Question# 7 Frequencies of Perceptions of the Directors of Special Services Toward the Effectiveness of the Alternate Proficiency Assessment since the Implementation of New Jersey's State Assessment System for Students with Severe Disabilities

The average response to the PASE survey ranged from an average low score of 1.25 to

a high score of 3.25. The breakdown of frequencies is shown in Table 16, clustered as

follows: disagree with value 2 (66), neutral with a value of 3 (1), and agree with values of

4-5(0). Responses (98.5%) fell in the response area of disagree.

Table 16

Summary of Frequencies with Respect to the Effectiveness of the Alternate Proficiency Assessment

	Frequency	Percentage
Disagree	66	98
Neutral	1	2
Agree	0	0

The average score for all the PASE survey scores was 2.21 suggesting that

effectiveness of the Alternate Proficiency Assessment (APA) is minimal since the

implementation of New Jersey's state assessment system for students with severe disabilities. Itemized analysis of the survey questions 8, 11, 12 indicates a neutral response with an, average score of 3, that APA scores may distort the districts adequate yearly progress (AYP); that the APA scores accurately reflect student progress toward the New Jersey Core Curriculum Content Standards (NJCCCS); and that the APA measures program quality more than student progress.

Research Question #8 Frequencies of Perceptions of the Directors of Special Services Toward the Effect Inclusion Since the Implementation of the New Jersey's State Assessment System

The average response to the PASE survey ranged from an average low score of 1 to a high score of 5. The breakdown of frequencies with respect to the effectiveness of inclusion in testing is shown in Table 17, clustered as follows: disagree with value 2 (22), neutral with a value of 3 (10), and agree with values of 4-5(35). Responses (52%) fell in the agree response area.

#### Table 17

Summary of Frequencies with Respect to the Effectiveness of Inclusion in Testing

	Frequency	Percentage
Disagree	22	33
Neutral	10	15
Agree	35	52

The average score for all the PASE survey scores was 3.20, suggesting that the New Jersey's state assessment system has had a somewhat positive effect on the inclusion of students with disabilities in general education classes.

#### Summary of Quantitative Research Results

Quantitative data results indicate that directors of special services:

- 1. Perceive assessment to have a positive effect on special education services;
- 2. Indicate that student achievement and learning has improved since the inclusion of students with disabilities in the New Jersey Assessment process;
- Indicate that professional development opportunities for special education teachers and general education teachers has improved since the inclusion of students with disabilities in the New Jersey Assessment process; and
- 4. Indicate that there was no significant difference between responses based on the district factor group, district enrollment, or the years of in the position of director of special services.

#### Qualitative Research Results

A focus group was held as part of the qualitative design for this study. The focus group was employed in an effort to provide quality information to New Jersey's school districts regarding the consequences of standardized assessment on students with disabilities. It is believed that the introduction of this technique would enhance the understanding of the quantitative survey data.

The questions for the focus group were developed by the researcher and determined and sequenced based on an analysis of the literature, focus group questions from previous doctoral dissertations, and based on open-ended survey questions from previous National Council for Educational Outcome research (Attachment B). The question guide was arranged in a logical manner to facilitate group discussion, and each type of question was designed for a specific purpose. A second jury of experts was asked to review the proposed questions for the focus group. Four directors of special education from preschool through Grade 12 districts were asked to determine that the questions demonstrated face and content validity. Questions were open-ended and non-specific so that the participant was able to interpret the questions based on his or her own experiences and follow the principles of qualitative interviewing to assure construct validity. Participants had the opportunity to verify the summary statement posed at the end of the session.

The composition of the focus group was similar to the survey participants. Participants of the focus group were directors of special services who have direct contact with the members of IEP team, including parents, case managers, general education and special education teachers. The directors are also responsible for monitoring curriculum and instructional implementation and reporting and analyzing reports and scores. The focus group was comprised of participants from K-8 and K-12 districts. The group was homogeneous in nature with respect to certification, but heterogeneous with respect to years of experience in education, district enrollment and DFG group.

There were 12 questions:

1. What evidence of change have you observed and experienced since the implementation of the New Jersey Assessment system relative to students with disabilities?

2. What do you see as a positive result of the participation of students with disabilities in the assessment and accountability system?

3. What do you see as a negative consequence of the participation of students with disabilities in the assessment and accountability system?

4. An intended primary outcome of standards-based reform is improved student learning through the establishment of high standards for all students and assessment of student progress toward meeting these standards. How does the NJ Assessment system do this for students with disabilities?

5. What impact has the NJ Assessment program and the inclusion of students with disabilities had on child study teams and teachers in your districts?

6. What impact, if any, has the NJ Assessment program and the inclusion of students with disabilities had on the students?

7. What have the parents of your students said about the NJ assessment system?

8. What impact has the inclusion of students with disabilities in the NJ Assessment process had on your district's accountability status?

9. What approach does your district use to determine which students participate in the alternate assessment?

10. How are accommodations determined in your district?

11. What are some of the major obstacles to including students with disabilities in the district accountability system?

12. Describe any other methods of assessment you believe would meet the assessment requirements?

The research questions became the basis for analyzing the qualitative data, with the focus of the questioning identifying accountability.

Comments of the directors of special services were codes according to themes and patterns. The researcher's professional judgment was used in the coding based on an analysis of words used in the focus groups (word repetitions, key-words in context, and professional terms). The comments were coded based on the directors of special services perception of accountability, student learning, and professional development as having a positive, neutral, or negative effect. Table 18 encapsulates this information based on the 12 questions posed during the focus group session.

# Table 18

Summary of Focus Group Responses

	Positive effect	Neutral effect	Negative effect	No comment	
Student Learning and Instructional Delivery	10 3	2		4	
Professional Development	8	4	3	1	
Effectiveness of APA for assessment purposes		4	10	2	
Participation in General Testing	4	6	1	5	
Effect of assessment on District AYP		10	3	3	
Accommodations	5	7		4	
Parent Reaction		5	6	5	
Impact on Inclusion	3	5		8	

a produced strength descent desce been accorded to the operation for provide billion we.

The comments made during the focus group session highlighted the overall perception of accountability of special education students. The directors often referenced specific concerns that existed within their district. Typically these concerns were related to federal or state government regulations and differences in philosophical opinion.

General comments and observations reflect the greatest impact of accountability on special education services is in the areas of instruction, student learning, and professional development. Directors of Special Services report:

1. there is an enhanced relationship between instruction in special education classes and the New Jersey Core Curriculum Content Standards (NJCCCS);

2. there is a decrease in altering the instructional materials; instead the learning expectations are being modified, but the same materials are used during instruction for regular education and special education;

3. replacement classes have been added to the curriculum to provide tutoring for students scoring not proficient or in danger of failing the state tests;

4. increased awareness of the importance of using accommodations throughout the day to day instruction and not just during testing situations;

5. instructional time in self-contained special education classes is compromised during the final weeks of the Alternate Proficiency Assessment (APA) to allow for preparation of the portfolio;

6. there has been an increase in professional development opportunities for general education teachers to include training on inclusion and instructional strategies and techniques, and;

7. allowance of professional development days for gathering of information and

compiling the portfolio materials for the APA

A second major theme for comments and observations focused on the APA and its impact on special education services. Directors of Special Services report:

1. it is difficult to establish consistency of testing procedures from year to year due to teacher turn-over;

2. it is difficulty to conduct item analysis to assess successes and failures due to the subjectivity of the assessment;

3. APA is not a comprehensive assessment of student progress since it only assesses three curricular areas and four goals from each area; and

4. although the purpose of the assessment is to improve student learning, the amount of time and attention invested in the preparation and presentation of the portfolio contradicts the purpose of the assessment.

#### Summar y

Data gathered in this study used both quantitative and qualitative analysis to determine the directors of special services perceptions of the consequences of high-stakes assessment and its impact on the increased demand for accountability on special education services. The data presented in this chapter illustrated that directors of special services do perceive assessment to have a positive effect on special education services. Specifically student achievement and learning and professional development within the district has improved since the inclusion of students with disabilities in the New Jersey Assessment process. There was no significant difference between responses based on the district factor group, district enrollment, or the years of in the position of director of special services. The qualitative data generated by the focus group provided a broad, comprehensive view of the directors' perceptions of accountability and its impact on special education services. Student achievement and instructional delivery and professional development are indicated to have been positively affected while effectiveness of the APA to assess student progress relative to the NJCCCS are indicated to have had a negative impact. These findings expand the research in the area of assessment and students with disabilities and provided a compilation of research that can be further explored within the context of accountability.

Chapter V

#### Summary

Knowing the extent to which all students are learning, is used to define educational policy, district funding, curricular decisions, and student class placement. As a means to assess progress, states have implemented large-scale, high-stakes testing. Local school districts are held accountable for student progress by the reporting of test scores.

Federal policy and the quest for improved schools have elevated the importance of assessment results for the purpose of school improvement and student achievement for all students, including students with disabilities. The IASA, IDEA '97, IDEA 2004 and NCLB 2001 are the most recent legislation to impact the educational reform movement that considers the learning needs of all students. Each of these laws has the common requirement that students with disabilities be included in the assessment system.

As a result of this legislation, the New Jersey Department of Education and local school districts had to confront what it means for all students to achieve at high standards. This indicates a policy shift away from the special education ideal of students meeting individualized goals to the inclusion of their progress into a standardized system of accountability. Directors of special services are faced with challenges of merging the ideals of special education in to the one size fits all general education system. They must lead their district in the policy shift to assure special education students are appropriately and accurately assessed.

- Studies have been cited in this dissertation of classroom teachers' perceptions of the alternate assessment process and of state directors of special education related to issues of accountability and assessment reliability and validity. This research study was designed to explore the consequences of large-scale high-stakes assessment for students with disabilities through the point of view, experiences and perceptions of the directors of special services in school districts in central and northern New Jersey. Directors of special services, being the primary administrator responsible for special education student placement and progress, have a valid insight into the usefulness of New Jersey's assessment system and its impact on special education services. The study attempted to collect and describe the directors' perceptions of the impact of the New Jersey Assessment System on special education services.

The methodology for the study included both quantitative and qualitative measures. To answer the research questions, a survey entitled "Perceptions of Accountability Special Education" (PASE) survey was created. A first draft survey was sent to a jury of experts who critiqued the contents and made suggestions. The survey was amended and sent to a pilot group for further refinement. Out of the 100 surveys sent, 67 useable surveys were returned.

Descriptive statistical data regarding overall directors of special services perceptions of accountability and its impact on special education services was provided by this study. The dependent variable was the average score on the PASE survey. Directors' perceptions were measured for the dependent variable based on like responses on the Likert scale instrument.

97

All three independent variables were categorical. They included District Factor Group which is an indicator of socioeconomic status, district enrollment, and years of experience in the position of director of special services.

Three null hypothesis were developed to guide the statistical analysis of the study. These included:

1. There is no significant difference in the average perception of accountability score is the same for all three district socioeconomic status groups.

2. There is no significant difference in the average perception of accountability score is the same for all five district size groups.

3. There is no significant difference in the average perception of accountability score is the same for all four directors of special services experience level groups.

Research questions quantified the directors' perceptions concerning changes in student achievement and instruction, use of testing accommodations, changes in professional development, the effectiveness of the Alternate Proficiency Assessment (APA), and the effect of including students with disabilities in the New Jersey assessment process. Certain assumptions were inherent to this study. It was assumed that directors responded honestly to anonymous survey. Further it was assumed the focus group participants were honest in sharing their opinions of accountability and its impact on special education services.

As with any research study, limitations exist that may pose a threat to finding answers to the research questions. Survey research is typically strong on reliability but weak on validity. To enhance the validity, the survey was piloted with directors of special services. The research methods were appropriate for the intent of this study, but it is important to note that caution must be taken when interpreting and attempting to generalize the findings on a state level or between states. This study was not designed to show causal relationships between the consequences of large-scale high-stakes assessment and specific educational outcomes for students with disabilities.

A qualitative research methodology was employed to help control limitations and the inability of the directors to provide open-ended responses. A focus group was used in an attempt to place the survey results within the proper context. However, it must be noted that focus group participants represented a sample population of the directors of special services in central and northern New Jersey.

#### Findings

The data analysis revealed that directors of special services do perceive assessment and the need to be held accountable for student progress to have a neutral to positive effect on special education services. Specifically student achievement and professional development within the district has improved since the inclusion of students with disabilities in the New Jersey Assessment process.

With respect to student achievement and learning, directors perceived an improvement in the special education teachers' knowledge of the general education curriculum since the inclusion of students with disabilities in the New Jersey Assessment system. Focus group participants commented that there is an enhanced relationship between instruction in special education classes and the New Jersey Core Curriculum Content Standards (NJCCCS) and a decrease in the altering of the instructional materials. Instead the learning expectations are being modified, but the same materials are used during instruction for regular education and special education. However caution should be used when determining if the assessment process has had a direct effect on student achievement since the study was not designed to show a causal relationship between the consequences of high-stakes testing and student achievement.

Another perceived benefit of the accountability mandate is the development of replacement classes, which have been added to the curriculum to provide tutoring for students scoring not proficient or in danger of failing the state tests.

Survey results indicate there was no change in the curriculum modifications by the general education teachers' when students with disabilities are included in their classes. Since the modifications are guided by the IEP they were consistently implemented. The survey and focus group participants were in agreement that curricular accommodations are consistently used for instruction and testing since the implementation of the New Jersey's state assessment system. An increased awareness of the importance of using accommodations throughout the day-to-day instruction and not just during testing situations has become evident with teachers since the inclusion of students with disabilities in the assessment process.

Directors also perceived that students with disabilities scores accurately reflect progress toward the NJCCCS when taking the general education assessment and the Alternate Proficiency Assessment (APA).

Directors also perceive that another assessment system other than the one that is currently being used would more accurately assess student progress for students with disabilities. General comments and observations reflect the greatest impact of
accountability on special education services is in the areas of instruction, student learning, and professional development.

There was agreement in the finding that effectiveness of the Alternate Proficiency Assessment (APA) is minimal since the implementation of New Jersey's state assessment system for students with severe disabilities. Survey results and focus group responses indicate that APA scores distort the district's Adequate Yearly Progress (AYP) and that the APA measures program quality more than student progress. Survey participants and focus group participants indicated a positive effect of the APA is that the APA scores accurately reflect student progress toward the NJCCCS However, the focus group participants stated that the APA is limiting in noting student progress. This is due to the assessment being based on the teacher selection to assess the student on two language arts goals, two math goals, and three science goals. The directors stated the assessment does not accurately reflect the overall student learning as compared to the comprehensive general assessment.

Students are being included in regular education classes more frequently than before the assessment mandate. Caution should be used, however, when concluding that the assessment mandate has resulted in more students with disabilities being included in regular classrooms. A factor influencing this movement may be the mandate for inclusion and a move away from self-contained classes.

Survey respondents and focus group participants perceived an improvement in professional development. Data from the focus group indicates that there has been an increase in professional development opportunities for general education teachers to include training on inclusion and instructional strategies and techniques. There is an increased awareness of the NJCCCS for special education teachers.

The hypotheses developed for the research were supported by the data. Specifically the hypothesis that the average perception of accountability score is the same for all 3 district socioeconomic status groups was accepted implying that the socioeconomic group as represented by the DFG Grouping does not have a significant effect on perception of accountability as indicated by the average PASE score.

The remaining hypotheses were not supported by the data. The district size or years of experience of the director of special services does not have a significant effect on the perception of accountability as indicated by the average PASE score.

# Implications

The findings of this study, dealing with the directors of special services perceived impact of accountability on special education services has corroborated the findings in the literature: The impact of accountability in special education has improved the instructional process for students with disabilities. School districts are now held accountable to assure students with disabilities are held to the same academic standards as all other students resulting in higher expectations for the students with disabilities. There is an increased participation in assessment and accountability measures and there is improved instruction. Professional development opportunities have expanded for general education teachers to include training on inclusion and instructional strategies and techniques for students with disabilities. It is important to remember, however, that the inclusion of students with disabilities in high-stakes large-scale assessment is in its infancy and there are numerous pathways of inquiry that need to be considered. The recommendations made here will be limited to major summary points.

The inclusion of students with disabilities is viewed with mixed results. While it appears unlikely that educators and parents would argue that in the past that there have been low achievement expectations for students with disabilities, we still do not know the full impact of including students in high-stakes large-scale testing. The results of this study indicate some positive findings and some areas that continue to be of great concern. As found in the literature, systematic concerns related to the reliability and validity of the alternate assessment process exist. Questions remain on whether assessment meets the desired outcome of measuring student performance. Results indicted that the directors believe the assessment measures program quality more than student achievement. Directors also believed that instructional time is compromised due to the large amount of time allocated for the preparation of the portfolio.

# **Recommendations**

# Future Research

Future research should involve the collection of longitudinal data to ascertain if there is an association between the inclusion of students in high-stakes large-scale testing and student achievement as evidenced by meeting state standards. Research should investigate the effects of students with disabilities being required to achieve at the same grade level standards as non-disabled students. Accountability systems should be examined to look at the impact of student scores on planning and preparation, the classroom environment, instruction, and professional development.

Further research should be conducted on the use of the alternate assessment. Rigorous standards must be established that assure the validity and reliability of the alternate assessment process. The assessment must be developed or refined to accurately reflect student progress in all areas not just a few selected goals. The assessment process must become more objective and comprehensive in its evaluation of student performance. While it is important to assess program quality and performance more weight should be placed on student achievement that the program components. The process must also become user friendly and time efficient. Until these criteria can be established, questions will continue to remain about the appropriateness and benefits of the alternate assessment process.

Although the directors of special services have direct knowledge of policy and curriculum, their perceptions may not accurately reflect the implementation concerns related to the assessment process. Therefore, conducting future research with classroom personnel, specifically the classroom teachers, may be helpful when comparing results related to improvement in instruction and student learning.

Future surveys should include additional demographic criteria. Understanding the types of disabilities served within the director of special services district would provide additional data on the effects of the assessment process relative to the type of disability.

## State and National Policy Implications

Students' inclusion in the accountability process helps to assure their successful attainment of higher educational standards. The practice of using assessment scores for students with disabilities when determining school resources and rewards and sanctions needs to be examined in light of the need for equity, consistency, and effectiveness for schools with a diverse student population. The question remains, should students with disabilities testing results have the same impact on the final AYP as the performance of other students.

Included in the efforts to improve outcomes for students with disabilities is professional development. States and local districts must support the teachers in developing a high quality assessment of skills that is authentic if the alternate assessment if going to improve the outcomes of students with disabilities. Professional development opportunities should be examined to assure teachers have a clear understanding of the content for which the students are being held accountable. Sustainability factors for a quality professional development plan include a deliberate plan for training, coaching and modeling, opportunities for teachers to understand how the assessment fits their students, the availability of peer networks and support and collegial learning opportunities, and administrative support.

Pre-service programs should be examined to ensure strong curricular content knowledge by the graduates with degrees in special education. Reorganization of undergraduate coursework may have to be considered to assure the graduates are prepared for the demands of instruction in special education. Content standards that are restricted to academics and do not include a wide range of curricular options for students with disabilities should be examined. If the primary goal of the standards is to provide instruction for students that will prepare them for full participation in the community as an adult, then the standards need to reflect this goal for all students, not just students along an academic path. There is a critical need to design standards and assess students in a rigorous manner but in a manner that is applicable for all learners.

Reporting of assessment performance data is critical to true educational reform. States and local districts must establish a process for sharing the data and then use the information to improve outcomes and achievement level for all students. States should provide data from all test takers whether they participate with or without accommodations or use an alternate assessment. Records should be maintained on the use of accommodations during the test and on a day-to-day basis to assure consistency of use. The reporting process must be clear, concise, and readily understood if results are to be interpreted and useful. The data should be consistent between states to provide comparative information about schools, districts, and states.

#### Summary

The results of the current study indicate some positive findings and some areas that continue to be of great concern. There is an increased participation in assessment and accountability measures and there is improved instruction. Professional development opportunities have expanded for general education teachers to include training on inclusion and instructional strategies and techniques for students with disabilities. As found in the literature, a systematic concern related to the reliability and validity of the alternate assessment process exists. Questions remain to the effectiveness of the Alternate Proficiency Assessment and its ability to assess student progress toward the New Jersey Core Curriculum Content Standards. Results indicated that the directors believe the assessment measures program quality more than student achievement. Directors also believed that instructional time is compromised due to the large amount of time allocated for the preparation of the portfolio. The use of assessment is a key element in educational reform. Longitudinal accountability research needs to be conducted to ascertain if the current assessment process is sustainable. Further research should be conducted to determine if the current perceived improvements in instructional and professional development will have a significant and lasting effect on student learning and instruction.

107

References

•

- Albrecht, S., & Joles, C. (Winter, 2003). Accountability and access to opportunity: Mutually exclusive tenets under a hi-stakes testing mandate. *Preventing School Failure*, 47(2), 86-91.
- Allbritten, D., Mainzer, R., & Ziegler, D. (Jan/Feb 2004). Will students with disabilities be scapegoats for school failures? *Teaching Exceptional Children*, 36(3), 74-75.
- Almond, P., Quenemoen, R., Olsen, K., Thurlow, M. (2000) Gray areas of assessment systems (Synthesis Report 32). Retrieved March 4, 2004, from University of Minnesota, National Center of Educational Outcomes. http://education.umn.edu/NCEO/OnlinePubs/Synthesis32
- Amerin, A. L., & Berliner, D.C. (2002). High-stakes testing, uncertainty, and students learning. *Educational Policy Analysis Archives*, 10(18), Retrieved March 16, 2004, from http://epaa.asu.edu/eppa/v10n18
- American Educational Research Association (2000, July). AERA positioning statement concerning high-stakes testing in Pre-K12 education. Retrieved March 5, 2004, from http://www.aera.net/about/policy/stakes.htm
- Anderson, G. (1999). Fundamentals of educational research. New York: The Palmer Press.

Babbie. E. (1999) The basics of social research. Belmont, CO: Wadsworth.

- Boyatzis, R. (1998). Transforming qualitative information: Thematic analysis and code development. Thousand Oaks, CA: Sage.
- Browder, D. M. (2001). Curriculum and assessment for students with moderate and severe disabilities. New York: The Guilford Press.
- Browder, D.M., Fallin, K., Davis, S., & Karvonen, M. (2003). Consideration of what may influence student outcomes on alternate assessment. *Education and Training in Developmental Disabilities, 38*, 255-270.
- Browder, D.M., Spooner, F., Algozzine, R., Ahlgrim-Delzell, L.,& Karvonen, M. (2003). What we know and need to know about alternate assessment. *Exceptional Children*, 70 (1), 45-61.
- Chudowsky, N., & Behuniak, P. (1998, Winter). Using focus groups to examine the consequential aspect of validity. *Educational Measurement: Issues and Practice*. 17(4), 28-38.
- Cohen, M., & Heumann, J. (2001, January 12). Clarification on the role of the IEP even selecting individual accommodations, modifications in administration, and alternate assessments for state and district-wide assessment of student achievement.
   Memorandum from the United States Department of Education.
- Coleman, A., Palmer S. (January 2004). Review of final NCLB 2001 rules for severely disabled students. *Special Education Report*, 6-8.

Educational Policy Research Reform Institute (EPRRI), (Spring 2002). Policy updates, issue one. College Park: University of Maryland, Educational Policy Research Reform Institute, The Institute for the Study of Exceptional Children and Youth. Retrieved 11-1-04, from www.eprri.org

Elliott, J.L., Erickson, R.N., Thurlow, M., Shriner, J. (2000). State-level accountability for performance of students with disabilities: Five years of change. *The Journal of Special Education*, 34(1), p. 39-47.

Fink, A. (1995). The survey handbook. Thousand Oaks, CA: Sage.

Fowler, F.J. (1993). Survey research methods (2<sup>nd</sup> ed.). Newbury Park, CA: Sage.

Fraenkel, J.R. (1996) How to design research in education. New York: McGraw-Hill.

Gersten, R., Chard, D., & Baker, S. (2000). Factors enhancing sustained use of researchbased instruction practices. *Journal of Learning Disabilities*. 33(5), 445-447.

Improving America's Schools Act (IASA) 1994. Retrieved March 25, 2004 from http://www.ed.gov/legislation/ESEA/index.html

Individuals with Disabilities Act (IDEA), P.L. 102-119, 20 U.S.C. 1400et.seq. (1997). Retrieved March 25, 2004 from http://www.ed.gov/offices/OSERS/Policy/IDEA/the law.html

- Johnson, E., & Arnold, N. (2004). Validating an alternate assessment. Remedial and Special Education, 25(5), 266-275. Retrieved January 18, 2005, from http://web3.epnet.com/externalframe
- Kampfer, S. H., Horvath, L. S., Kleinert, H., L., & Kearns, J., F. (2001). Teachers' perceptions of one state's alternate assessment: Implications for practice and preparation. *Exceptional Children*, 67, 361-374.
- Kleinert, H.L., & Kearns, J. F. (2001). Alternate assessment: Measuring outcomes and supports for students with disabilities. Baltimore: Paul H. Brookes Publishing.
- Kleinert, H.L., Kennedy, S., & Kearns, J. F. (1999). The impact of alternate assessment: a statewide teacher's survey. *The Journal of Special Education*, 33, 93-102.
- Koretz, D., Stecher, B., Klein, S., & McCaffrey, D. (1994). The Vermont Portfolio Assessment Program: Findings and implications. *Educational Measurement: Issues* and Practice, 13 (3), 5-16.
- Krueger, R. (2000). Focus groups: A practical guide for applied research. Thousand Oaks, CA: Sage.
- Lane, S., Parke, C.S. & Stone, C.A. (1998, Summer). A framework for evaluating the consequences of assessment programs. *Educational Measurement: Issues and Practice*, 24-28.

Langenfeld, K., Thurlow, M., Scott, D. (1997, January). High-stakes testing for students: unanswered questions and implications for students with disabilities. National Center on Educational Outcomes. Retrieved March 18, 2004, from University of Minnesota, National Center of Educational Outcomes.

http://education.umn.edu/NCEO/OnlinePubs/NCLB 2001 disabilties.htm LaMorte, M. W. (2002) School law. (7<sup>th</sup> ed.). Boston: Allyn and Bacon.

Lawmakers Laud Final Regulations for Testing Disabled. (January 2004). Education USA, 8.

- Leddy, P.D. (1997) Practical research planning and design. (6<sup>th</sup> ed.). Upper Saddle River, NJ: Prentice Hall.
- Lehr, C., & Thurlow, M. (2003, October). Putting it all together: Including students with disabilities in assessment and accountability systems. National Center on Educational Outcomes. Retrieved April 22, 2004, from University of Minnesota, National Center of Educational Outcomes
  http://education.umn.edu/nceo/OnlinePubs/Policy16.htm
- Mann, F. (1970). Human relation skills in social research. In W.J. Filstead, (Ed.).
   Qualitative Sociology: Firsthand Involvement with the Social World. Chicago:
   Markham Publishing.
- Meherns, W. (1998). Consequences of assessment: What is the evidence? Educational Policy Analysis Archives 6, 13.

Messick, S. (1989). Validity. In R.L.Linn (Ed.) Educational Measurement (3<sup>rd</sup> ed.), New York: American Council on Education.

Miller, M.D., & Linn, R.L. (2000). Validation of performance-based assessments. Applied Psychological Measurement, 2, 367-378.

Murphy, J. (1980). Getting the facts: A fieldwork guide for evaluators and policy analysis. Santa Monica, CA: Goodyear.

New Jersey Alternate Proficiency Assessment 2004: Executive Summary. Retrieved February 1, 2006, from

http://www.nj.gov/njded/assessment/apa/2004apa\_executives\_summary.pdf

No Child Left Behind Act of 2001, 107 U.S.C., P.L. 107-110 (2002). Retrieved February 18, 2004 from http://www.ed.gov/policy/elsec/leg/esea02/index.html

No Child Left Behind has Major Implications for Special Education. (2002, November/December). CEC Today, 9, 4.

Patton, M.Q. (1990). Qualitative evaluation and research methods (2<sup>nd</sup> ed.) Newbury Park, CA: Sage.

President's Commission on Excellence in Special Education. (2002). A new era:
Revitalizing special education for children and families. Washington, DC: U.S.
Department of Education Office of Special Education and Rehabilitative Services.

- 115
- Quenemon, R., Rigney, S., & Thurlow, M. (2002). Use of alternate assessment results in reporting and accountability systems: Conditions for use based on research and practice (Synthesis Report 43). Minneapolis, MN: University of Minnesota, National Center of Educational Outcomes. Retrieved 1-11-04 from http://education.umn.edu/NCEO/OnlinePubs/Synthesis43
- Rea, L.M., & Parker, R.A. (1997). Designing and conducting survey research: A comprehensive guide. San Francisco, CA: Jossey-Bass.
- Roach, A. T., Elliott, S. N., Webb, N. L. (2005). Alignment of an alternate assessment with state academic standards: Evidence for the content validity of the Wisconsin alternate assessment. *The Journal of Special Education*, 38(4), 218-231.
- Rossi, P.H., Wright, J.D., & Anderson, A.B. (1983). Handbook of survey research. New York: Academic Press.

Salvia, J., & Ysseldyke, J.E. (2001). Assessment (8th edition). Boston: Houghton Mifflin.

Special Needs, Common Goals. (January 8, 2002). Education Week, 28(17), 7.

- Thompson, S., Quenemoen, R., Thurlow, M., Ysseldyke, J. (2003). Alternate assessments for students with disabilities. Thousand Oaks: Corwin Press, Inc.
- Thompson, S., & Thurlow, M. (2001). 2001 State special education outcomes: A report on state activities at the beginning of a new decade. Retrieved March 18, 2004 from University of Minnesota, National Center of Educational Outcomes

http://education.umn.edu/NCEO/OnlinePubs/htm

- Thurlow, M.L., Bielinski, J, Minnema, J., & Scott, J. (2002). Out-of-level testing revisited: New Concerns in the era of standards-based reform. In G. Tindal & T. M. Haladyna, (Eds.), Large-scale assessment programs for all students: Validity, technical adequacy, and implementation pp. 1-26. Mahwah, NJ: Erlbaum.
- Thurlow, M., & Bolt, S. (2001) Empirical support for accommodations most often allowed in state policy (Synthesis Report 41). Retrieved March 4, 2004, from University of Minnesota, National Center for Educational Outcomes http://education.umn.edu/NCEO/OnlinePubs/Synthesis41.html
- Thurlow, M., Elliott, J., & Ysseldyke, J. (1999). Out of level testing: Pros and cons (NCEO Policy Directions). Retrieved March 4, 2004 from University of Minnesota, National Center for Educational Outcomes http://education.umn.edu/NCEO/OnlinePubs/
- Thurlow, M. Elliott, J., Ysseldyke, J. (2003). Testing Students with Disabilities. Thousand Oaks: Corwin Press, Inc.
- Thurlow, M., & Minnema, J. (2001). States out-of-level testing policies (out of level testing report 4) Retrieved March 4, 2004 from University of Minnesota, National Center for Educational Outcomes http://education.umn.edu/NCEO/OnlinePubs/

00% fism http:///official/addition/page

- Thurlow, M., Wiley, H.I., Bielinski, J. (2003, April). Going public: What 2000-2001
  reports tell us about the performance of students with disabilities. Retrieved March 18, 2004, from University of Minnesota, National Center for Educational Outcomes http://education.umn.edu/NCEO/OnlinePubs/NCLB 2001 disabilities.htm
- Turner, M., Baldwin, L., Kleinert, H., & Kearns, J. (2001). An examination of the concurrent validity of Kentucky's alternate assessment system. The Journal of Special Education, 34, 69-76.
- U.S. Department of Education. (2002). The No Child Left Behind Act of 2001: Executive summary. Retrieved February 11, 2004, from http://www.ed.gov/NCLB 2001/overview/intro/execsumm.html
- U.S. Department of Education. (2003). No Child Left Behind. Retrieved February 11, 2004, from http://NoChildLeftBehind.gov/
- U.S. Department of Education. (2003). No Child Left Behind provision gives schools new flexibility and ensures accountability for students with disabilities-Fact sheet. Retrieved March 25, 2004 from http://www.ed.gov/NCLB 2001/freedom/local/specedfactsheet.html?exp=0
- U.S. Department of Education. (2003). Title I--Improving the Academic Achievement of the Disadvantaged. 34 CFR Part 200. Retrieved March 25, 2004, from http://www.ed.gov/legislation/FedRegister/finrule/2003-4/

- U.S. Department of Education, Office for Civil Rights. (2000). The use of tests as part of high-stakes decision-taking for students goal in a resource guide for educators and policymakers. Washington, DC: U.S.D.O.E.
  - Ysseldyke, J., & Olsen, K. (1997). Putting alternate assessment into practice: what to measure and possible sources of data. Retrieved March 4, 2004 from University of Minnesota, National Center for Educational Outcomes. http://education.umn.edu/NCEO/OnlinePubs/

# APPENDIX A

Perceptions of Accountability in Special Education

# Perceptions of Accountability in Special Education of Directors/Supervisors of Special Education

**Directions:** Please answer each question so that each response best reflects your attitude toward the New Jersey Assessment and Accountability System and its impact on students with disabilities.

Demographic Information

1. District enrollment as of October 15, 2004 ASSA Report (all not just special

education):		8		
Strongly Disagree	gr	Wentynl	gr	Strongly Agree
500-800				
<b>8</b> 01-1000				
1001-3000				
<b>3001-5000</b>				
More than 5001				
2. District DFG Group:				
□A,B				
CD, DE, FG, GH, H				
□I,J				

3. Number of years served as director/supervisor of special education (in your present district or any other district):

Less than 1 year

1-3 years 4-9 years

 $\square$  10 or more years

If less than 3 years, what was your previous certificated position

District Assessment and Performance

1. I believe the NJ assessment system has had a positive affect on the inclusion of students with disabilities in eneral education classes.

1	2	3	4	5
	Disa ee	Neutral	A ee	

Strongly DisagreegrgrgrStrongly Disagree2. glabelieve the inclusiony of students with disabilities in the NJ Assessment systemproved the general education teachers' curricular knowledge and ability to reconstruct the general education teachers' curricular knowledge and ability to reconstruct the general education teachers' curricular knowledge and ability to reconstruct the inclusion of students with disabilities in the NJ Assessment infarrometal Ifficagree ia education teachers' knowled e of the energial education teachers' knowled education teachers' knowled e of the energial education teachers' knowled educatio	04	2			
2. gheelieve the inclusion of students with disabilities in the NJ Assessment system improved the general education teachers' curricular knowledge and ability to re- control the general education teachers' curricular knowledge and ability to re- control the general education teachers' curricular knowledge and ability to re- control the general education teachers' curricular knowledge and ability to re- a the inclusion of students with disabilities in the NJ Assessment instronged the inclusion of students with disabilities in the NJ Assessment instronged the inclusion of students with disabilities in the NJ Assessment instronged the inclusion of students with disabilities in the NJ Assessment instronged the inclusion of students with disabilities in the NJ Assessment instronged the energial education teachers' knowled e of the energial education teachers' for the energial education teachers' knowled e of the energial education teachers' for the energial education teachers' knowled e of the energial education teachers' for the energial education teachers' knowled e of the energial education teachers' for the energial education teachers' knowled e of the energial education teachers' for the energial education teachers' knowled e of the energial education teachers' for the energial education teachers' knowled e energial education teachers' knowled	Strongly Disagree	gr		gr	Strongly Ag
In proved the general education contents of the field where $f$ is a fi	. helieve the inclus	ion of students v	with disabilities in	the NJ Assess	nent system has
Calloring y initiagreegrgrgrgrstr1234Disa eeNeutralA eey3In my view the inclusion of students with disabilities in the NJ Assessment instronged Illicagreeia education teachers' knowled e of the energial education to the energial education teachers' knowled e of the energial education to the energial education at education at education at education at education to the energial education at educat	Arrientur Discorrect			wieuge and abi	Strengly A
Image: DisabilitiesDisabilitiesNeutralA eeyy3. In my view the inclusion of students with disabilities in the NJ Assessmentinstronged Ibinagreeiaeducation teachers' knowled e of the energal education teachers' knowled end teachers' knowled end teachers' knowled education teachers' knowled end teachers' knowled education teachers' knowled end teachers' teachers' knowled end teachers' knowled end teachers' knowled end teachers' teachers' knowled end teachers' teachers' knowled end teachers' knowled end teachers' knowl	1	2	3		Strongly Ag
Y3. In my view the inclusion of students with disabilities in the NJ Assessment instronged Itheoretical education teachers' knowled e of the eneral education teachers' knowled	We make without West Party	Disa ee	Neutral	A ee	at source for
3. In my view the inclusion of students with disabilities in the NJ Assessment instronged Itheoretical education teachers' knowled e of the energial education teachers' knowled education teachers' knowled education teachers' knowled education teachers' knowled education teachers with disabilities of the general assessments stem. 1 2 3 4 Disa ee Neutral A ee 8. In my view the inclusion of the Alternate Proficiency Assessment test score students with disabilities distorts m district's AYP results. 1 2 3 4 Disa ee Neutral A ree 9. I believe the scores for students with disabilities on the general education as accuratel reflect student ro ress toward the CCCS.	Mentio solo Dissidut	у			
instronged Itheographical education teachers' knowled e of the energial education teachers with disabilities distorts m district's AYP results. <ul> <li>1</li> <li>2</li> <li>3</li> <li>4</li> <li>1</li> <li>2</li> <li>3</li> <li>4</li> <li>1</li> <li>2</li> <li>3</li> <li>4</li> <li>2</li> <li>3</li> <li>4</li> <li>4</li> <li>4</li> <li>6</li> </ul>	. In my view the incl	lusion of studen	ts with disabilities	in the NJ Asse	ssment system l
1234DisaeeNeutralAeeypggrStrongly DisagreegrgrStrongly Disagreegrgrgrstrongly Disagreeypgrgrgrstrongly DisagreeStrongly Disagreegrgrgrstrongly DisagreegrgrgrgrgrADisaeeNeutralAee5. In my view the curricular accommodations are consistently used for instructestin .1234is aeeDisaeeNeutralAee6. I believe the inclusion of the general assessment test scores for students with disabilities distorts mdistrict's AYP results.1234DisaeeNeutralAee7. In my view there are more positive changes occurring in my district concerninclusion of all students than negative changes, including those with the most score itive disabilities, in our districts assessments stem.1234DisaeeNeutralAee8. In my view the inclusion of the Alternate Proficiency Assessment test score students with disabilities distorts mdistrict's AYP results.1234DisaeeNeutralAee9. I believe the scores for students with disabilities on the general education as accuratel reflect student ro ress toward the CCCS.44	Stronged Bisagreeial	ducation teache	rs' knowled e of	the eneral edu	cationrongiculg
ypgStrongly DisagreegrgrStrongly DisagreegrgrgrStrongly DisagreeDisaeeNeutralADisaeeNeutralA1234eeDisaeeNeutral1234eeDisaeeNeutral1234eeDisaeeNeutral1234DisaeeNeutralA1234DisaeeNeutralA1234DisaeeNeutralA1234DisaeeNeutralA1234DisaeeNeutralA1234DisaeeNeutralA1234DisaeeNeutralA1234DisaeeNeutralA1234DisaeeNeutralA1234DisaeeNeutralA12340014141234001414123400141412 <td>1</td> <td>2 Disa ee</td> <td>3 Neutral</td> <td>4 A ee</td> <td>5</td>	1	2 Disa ee	3 Neutral	4 A ee	5
Strongly DisagreegrgrgrStrongly DisagreeDisa eeNeutralA ee $A$ Disa eeNeutralA ee1234eeDisa eeNeutralA ee6. I believe the inclusion of the general assessment test scores for students with disabilities distorts m district's AYP results.A1234Disa eeNeutralA ee7. In my view there are more positive changes occurring in my district concern inclusion of all students than negative changes, including those with the most score itive disabilities, in our districts assessments stem.A1234Disa eeNeutralA ee8. In my view the inclusion of the Alternate Proficiency Assessment test score students with disabilities distorts m district's AYP results.A1234Disa eeNeutralA ee8. In my view the inclusion of the Alternate Proficiency Assessment test score students with disabilities distorts m district's AYP results.A1234Disa eeNeutralA ee9. I believe the scores for students with disabilities on the general education as accuratel reflect student ro ress toward the CCCS.A	у	pg	and the Lands.		
Strongly Disagreegrgrstrongly disagreeDisa eeNeutralA ee $A$ Disa eeNeutralA eeS. In my view the curricular accommodations are consistently used for instructestin .121234eeDisa eeNeutralA ee6. I believe the inclusion of the general assessment test scores for students with disabilities distorts m district's AYP results.121234Disa eeNeutralA ee7. In my view there are more positive changes occurring in my district concerninclusion of all students than negative changes, including those with the most score itive disabilities, in our districts assessments stem.11234Disa eeNeutralA ee8. In my view the inclusion of the Alternate Proficiency Assessment test score students with disabilities distorts m district's AYP results.11234Disa eeNeutralA ree9. I believe the scores for students with disabilities on the general education as accuratel reflect student ro ress toward the CCCS.					
Disa eeNeutralAee5. In my view the curricular accommodations are consistently used for instructestin1234eeDisa eeNeutralAeeDisa eeNeutralAeeDisa ee1234Disa eeNeutralAee7. In my view there are more positive changes occurring in my district concerninclusion of all students than negative changes, including those with the most second time disabilities, in our districts assessments stem.123123234Disa eeNeutralA ee8. In my view the inclusion of the Alternate Proficiency Assessment test score students with disabilities distorts m district's AYP results.123234Disa eeNeutralAee	Strongly Disagree	gr	yotamin	gr A	Strongly Ag
5. In my view the curricular accommodations are consistently used for instructestin . 1 2 3 4 ee Disa ee Neutral A ee 6. I believe the inclusion of the general assessment test scores for students with disabilities distorts m district's AYP results. 1 2 3 4 Disa ee Neutral A ee 7. In my view there are more positive changes occurring in my district concerning inclusion of all students than negative changes, including those with the most score itive disabilities, in our districts assessments stem. 1 2 3 4 Disa ee Neutral A ee 8. In my view the inclusion of the Alternate Proficiency Assessment test score students with disabilities distorts m district's AYP results. 1 2 3 4 Disa ee Neutral A ee 8. In my view the inclusion of the Alternate Proficiency Assessment test score students with disabilities distorts m district's AYP results. 1 2 3 4 Disa ee Neutral A ee 9. I believe the scores for students with disabilities on the general education as accuratel reflect student ro ress toward the CCCS.		Disa ee	Neutral	A ee	
<ul> <li>6. I believe the inclusion of the general assessment test scores for students with disabilities distorts m district's AYP results. <ol> <li>2</li> <li>3</li> <li>4</li> <li>Disa ee</li> <li>Neutral</li> <li>A ee</li> </ol> </li> <li>7. In my view there are more positive changes occurring in my district concern inclusion of all students than negative changes, including those with the most score itive disabilities, in our districts assessments stem. <ol> <li>2</li> <li>3</li> <li>4</li> <li>Disa ee</li> </ol> </li> <li>8. In my view the inclusion of the Alternate Proficiency Assessment test score students with disabilities distorts m district's AYP results. <ol> <li>2</li> <li>3</li> <li>4</li> <li>Disa ee</li> <li>Neutral</li> <li>A ee</li> </ol> </li> <li>9. I believe the scores for students with disabilities on the general education as accuratel reflect student ro ress toward the CCCS.</li> </ul>					
disabilities distorts m district's AYP results. 1 2 3 4 Disa ee Neutral A ee 7. In my view there are more positive changes occurring in my district concern inclusion of all students than negative changes, including those with the most s co itive disabilities, in our districts assessments stem. 1 2 3 4 Disa ee Neutral A ee 8. In my view the inclusion of the Alternate Proficiency Assessment test score students with disabilities distorts m district's AYP results. 1 2 3 4 Disa ee Neutral A ee 9. I believe the scores for students with disabilities on the general education as accuratel reflect student ro ress toward the CCCS.		ion of the conor			
1       2       3       4         Disa ee       Neutral       A ee         7. In my view there are more positive changes occurring in my district concerninclusion of all students than negative changes, including those with the most second time disabilities, in our districts assessments stem.         1       2       3       4         Disa ee       Neutral       A ee         8. In my view the inclusion of the Alternate Proficiency Assessment test score students with disabilities distorts m district's AYP results.       4         1       2       3       4         Disa ee       Neutral       A ee         9. I believe the scores for students with disabilities on the general education as accuratel reflect student ro ress toward the CCCS.       4	. I believe the inclus	ion of the genera	al assessment test	scores for stude	ents with
Disa eeNeutralA ee7. In my view there are more positive changes occurring in my district concern inclusion of all students than negative changes, including those with the most s co itive disabilities, in our districts assessments stem.1234Disa eeNeutralA ee8. In my view the inclusion of the Alternate Proficiency Assessment test score students with disabilities distorts m district's AYP results.1234Disa eeNeutralA ree9. I believe the scores for students with disabilities on the general education as accuratel reflect student ro ress toward the CCCS.	. I believe the inclus isabilities distorts m	district's AYP	al assessment test results.	scores for stude	ents with
<ul> <li>7. In my view there are more positive changes occurring in my district concern inclusion of all students than negative changes, including those with the most score itive disabilities, in our districts assessments stem. <ol> <li>2</li> <li>3</li> <li>4</li> <li>Disa ee</li> </ol> </li> <li>8. In my view the inclusion of the Alternate Proficiency Assessment test score students with disabilities distorts m district's AYP results. <ol> <li>2</li> <li>3</li> <li>4</li> </ol> </li> <li>9. I believe the scores for students with disabilities on the general education as accuratel reflect student ro ress toward the CCCS.</li> </ul>	. I believe the inclus isabilities distorts m 1	district's AYP 2	al assessment test results. 3	scores for stude	ents with 5
<ul> <li>inclusion of all students than negative changes, including those with the most score itive disabilities, in our districts assessments stem.</li> <li> <ol> <li>2</li> <li>3</li> <li>4</li> </ol> </li> <li>Disa ee Neutral A ee</li> </ul> 8. In my view the inclusion of the Alternate Proficiency Assessment test score students with disabilities distorts m district's AYP results. <ul> <li>1</li> <li>2</li> <li>3</li> <li>4</li> <li>Disa ee Neutral A ee</li> </ul> 9. I believe the scores for students with disabilities on the general education as accuratel reflect student ro ress toward the CCCS.	. I believe the inclus isabilities distorts m 1	district's AYP 2 Disa ee	al assessment test results. 3 Neutral	scores for stude 4 A ee	ents with 5
<ul> <li>a 1 2 3 4</li> <li>b 1 2 3 4</li> <li>b 2 3 4</li> <li>b 2 3 4</li> <li>b 2 3 4</li> <li>b 3 1 1 2 3 4</li> <li>c 1 3 4</li> <li>c 1 4 4</li> <lic 1="" 4="" 4<="" li=""> <li>c 1 4 4</li> <li>c 1 4 4</li> <li>c 1 4</li></lic></ul>	I believe the inclus isabilities distorts m 1 . In my view there as	district's AYP 2 Disa æ	al assessment test results. 3 Neutral changes occurrin	scores for stude 4 A ee g in my district	ents with 5 concerning the
1       2       3       4         Disa ee       Neutral       A ee         8. In my view the inclusion of the Alternate Proficiency Assessment test score students with disabilities distorts m district's AYP results.       1         1       2       3       4         Disa ee       Neutral       A ree         9. I believe the scores for students with disabilities on the general education as accuratel reflect student ro ress toward the CCCS.	I believe the inclus isabilities distorts m l . In my view there an iclusion of all studen	district's AYP 2 Disa ee re more positive ts than negative	al assessment test results. 3 Neutral changes occurrin changes, includin	scores for stude 4 A ee <u>g in my district</u> g those with the	ents with 5 <u>concerning</u> the e most significat
<ul> <li>8. In my view the inclusion of the Alternate Proficiency Assessment test score students with disabilities distorts m district's AYP results. <ol> <li>2</li> <li>3</li> <li>4</li> </ol> </li> <li>9. I believe the scores for students with disabilities on the general education as accuratel reflect student ro ress toward the CCCS.</li> </ul>	I believe the inclus isabilities distorts m l In my view there an iclusion of all studen o itive disabilities,	district's AYP 2 Disa ee re more positive ts than negative in our districts	al assessment test results. 3 Neutral changes occurrin changes, includin assessments ster	scores for stude 4 A ee <u>g in my district</u> g those with the n.	ents with 5 <u>concerning the</u> e most significa
<ul> <li>8. In my view the inclusion of the Alternate Proficiency Assessment test score students with disabilities distorts m district's AYP results. <ol> <li>2</li> <li>3</li> <li>4</li> </ol> </li> <li>9. I believe the scores for students with disabilities on the general education as accuratel reflect student ro ress toward the CCCS.</li> </ul>	<ul> <li>I believe the inclus isabilities distorts m <ol> <li>I</li> <li>In my view there an inclusion of all studen</li> <li>itive disabilities, itigation</li> </ol> </li> </ul>	district's AYP 2 Disa ee re more positive ts than negative in our districts 2 Disa e	al assessment test results. 3 Neutral changes occurrin changes, includin assessments ster 3	scores for stude 4 A ee <u>g in my district</u> ig those with the n. 4	ents with 5 <u>concerning the</u> e most significat 5
1       2       3       4         Disa ee       Neutral       A ree         9. I believe the scores for students with disabilities on the general education as accuratel reflect student ro ress toward the CCCS.	<ul> <li>I believe the inclus isabilities distorts m <ol> <li>I</li> </ol> </li> <li>In my view there an inclusion of all studen o itive disabilities, iso itive disabilities, itive disabilities,</li></ul>	district's AYP 2 Disa ee re more positive ts than negative in our districts 2 Disa ee	al assessment test results. Neutral changes occurrin changes, includin assessments ster 3 Neutral	scores for stude 4 A ee <u>g in my district</u> ig those with the n. 4 A ee	ents with 5 <u>concerning</u> the e most significat 5
Disa ee Neutral A ree 9. I believe the scores for students with disabilities on the general education as accuratel reflect student ro ress toward the CCCS.	<ul> <li>I believe the inclus isabilities distorts m <ol> <li>In my view there and the inclusion of all studen</li> <li>itive disabilities, inclusion is abilities, inclusion is abilities, inclusion is abilities, inclusion is abilities.</li> </ol> </li> </ul>	district's AYP 2 Disa ee re more positive ts than negative in our districts 2 Disa ee lusion of the Alte	al assessment test results. 3 Neutral changes occurrin changes, includin assessments ster 3 Neutral ernate Proficiency listrict's AYP resu	scores for study 4 A ee <u>g in my district</u> 1g those with the n. 4 A ee y Assessment tends.	ents with 5 <u>concerning the</u> e most significat 5 est scores for
9. I believe the scores for students with disabilities on the general education as accuratel reflect student ro ress toward the CCCS.	<ul> <li>I believe the inclus isabilities distorts m <ol> <li>In my view there an inclusion of all studen</li> <li>itive disabilities, il</li> </ol> </li> <li>In my view the includents with disabilities in the inclusion of all students with disabilities wi</li></ul>	district's AYP 2 Disa ee re more positive ts than negative in our districts 2 Disa ee lusion of the Alte ies distorts m d 2	al assessment test results. Neutral changes occurrin changes, includin assessments ster 3 Neutral ernate Proficiency listrict's AYP resu 3	scores for stude 4 A ee g in my district g those with the $A$ ee 4 A ee 4 A ee 4 A ee 4 4 4 4 4 4 4 4	ents with 5 <u>concerning</u> the e most significan 5 est scores for 5
accuratel reflect student ro ress toward the CCCS.	<ul> <li>I believe the inclus isabilities distorts m <ol> <li>In my view there and the inclusion of all studen</li> <li>itive disabilities, it is is is inclusion.</li> </ol> </li> <li>In my view the inclusion of all students with disabilities is inclusion.</li> </ul>	district's AYP 2 Disa ee re more positive ts than negative in our districts 2 Disa ee lusion of the Alte ies distorts m d 2 Disa ee	al assessment test results. 3 Neutral changes occurrin changes, includin assessments ster 3 Neutral ernate Proficiency listrict's AYP resu 3 Neutral	scores for stude 4 A ee g  in my district ig those with the A ee A see A seessment tends. 4 A ree	ents with 5 <u>concerning the</u> e most significat 5 est scores for 5
accurates reflect stations to ress toward the CCCS,	<ul> <li>I believe the inclus isabilities distorts m <ol> <li>In my view there an inclusion of all studen</li> <li>itive disabilities,</li> </ol> </li> <li>In my view the includents with disabilities</li> </ul>	district's AYP 2 Disa ee re more positive ts than negative in our districts 2 Disa ee lusion of the Alta ies distorts m d 2 Disa ee	al assessment test results. Neutral changes occurrin changes, includin assessments ster Neutral ernate Proficiency listrict's AYP resu Neutral	scores for stude 4 A ee g in my district g those with the A ee A see A A ee A A ee A A ee A A ree A ree	ents with 5 <u>concerning the</u> e most significat 5 est scores for 5
1 2 3 1	<ul> <li>I believe the inclus isabilities distorts m <ol> <li>In my view there an inclusion of all studen</li> <li>itive disabilities, il</li> </ol> </li> <li>In my view the includents with disabilities <ol> <li>I believe the scores</li> </ol> </li> </ul>	district's AYP 2 Disa ee re more positive ts than negative in our districts 2 Disa ee lusion of the Alte ies distorts m d 2 Disa ee for students wit	al assessment test results. Neutral changes occurrin changes, includin assessments ster Neutral ernate Proficiency listrict's AYP resu Neutral h disabilities on the	scores for stude 4 A ee g in my district g those with the A ee A see A see A see A see A ree he general educ	ents with 5 <u>concerning the</u> e most significat 5 est scores for 5 cation assessmer
I 2 J 4 Disa ee Neutral A ee	<ul> <li>I believe the inclus isabilities distorts m <ol> <li>In my view there an inclusion of all studen</li> <li>itive disabilities, i</li> </ol> </li> <li>In my view the includents with disabilities <ol> <li>I believe the scores couratel reflect studet</li> </ol> </li> </ul>	district's AYP 2 Disa ee re more positive ts than negative in our districts 2 Disa ee lusion of the Alta ies distorts m d 2 Disa ee for students wittent ro ress tow	al assessment test results. 3 Neutral changes occurrin changes, includin assessments ster 3 Neutral ernate Proficiency listrict's AYP resu 3 Neutral ch disabilities on the vard the CCCS.	scores for stude 4 A ee g in my district g those with the A ee A see A see A see A see A ree he general educe A	ents with 5 <u>concerning the</u> e most significat 5 est scores for 5 cation assessmen

		THE S OF SHOOM CO		
10 In mar evicence the	y p	rogr	tition in man lint	nidt home insurance
TO. In my view the	professional devel	opinent opportun	nues in my dist	not have improved
of students with dis	ng instruction for a	dized testin	re disabilities s	unce attounder and
			4	5
1	Z Disa aa	J Noutral	4	5
	Disa ce	incut al	A CC	
11 I believe the so	orec for students u	rith disabilities on	the Alternate P	roficiency
Assessment accurate	el reflect student	to ess toward f	be CCCS	Tonciency
1	7	3	4	5
1	Disa ee	Neutral	A ee	5
		. (output		
12 I believe that th	e Alternate Profici	iency Assessment	measures prog	ram quality more
12. I bolleve ulut ul		ency rissessment	measures prog	and quanty more
than student efform	nance			
than student erform	nance.	3	4	 - -
than student erform 1	nance. 2 Disa ee	3 Neutral	4 A ee	5
than student erform 1	nance. 2 Disa ee	3 Neutral	4 A ee	5 ea
than student erform 1 13. I believe the Ne	nance. 2 Disa ee w Jersey general a	3 Neutral	4 A ee NJ ASK 3-4, G	5 EPA, HSPA) hav
than student erform 1 13. I believe the Ne im roved instruction	nance. 2 Disa ee w Jersey general a nal	3 Neutral assessment tests ( district for s	4 A ee NJ ASK 3-4, G students with di	5 EPA, HSPA) hav sabilities.
than student erform 1 13. I believe the Ne im roved instruction 1	nance. 2 Disa ee ew Jersey general a nal	3 Neutral assessment tests ( district for s 3	4 A ee NJ ASK 3-4, G students with di 4	5 EPA, HSPA) hav sabilities. 5
than student erform 1 13. I believe the Ne im roved instruction 1 ee	nance. 2 Disa ee ew Jersey general a nal Disa ee	3 Neutral assessment tests ( district for s 3 Neutral	4 A ee NJ ASK 3-4, G students with di 4 A ee	5 EPA, HSPA) hav sabilities. 5 ee
than student erform 1 13. I believe the Ne im roved instruction 1 ee	nance. 2 Disa ee ew Jersey general a nal Disa ee	3 Neutral assessment tests ( district for s 3 Neutral	4 A ee NJ ASK 3-4, G students with di 4 A ee	5 EPA, HSPA) hav sabilities. 5 ee
than student erform 1 13. I believe the Ne im roved instruction 1 ee 14. In my view the	nance. 2 Disa ee ew Jersey general a nal Disa ee parents of my stud	3 Neutral assessment tests ( district for s 3 Neutral lents with disabili	4 A ee NJ ASK 3-4, G students with di 4 A ee ties believe the	5 EPA, HSPA) hav sabilities. 5 ea New Jersey
than student erform 1 13. I believe the Ne im roved instruction 1 ee 14. In my view the Alternate Proficienc	nance. 2 Disa ee ew Jersey general a nal Disa ee parents of my stud Assessment has	3 Neutral assessment tests ( district for s 3 Neutral lents with disabili im roved the sub	4 A ee NJ ASK 3-4, G students with di 4 A ee ties believe the iect content the	5 EPA, HSPA) hav sa bilities. 5 ea New Jersey ir child is learnin
than student erform 1 13. I believe the Ne im roved instruction 1 ee 14. In my view the Alternate Proficienc 1	nance. 2 Disa ee ew Jersey general a nal Disa ee parents of my stud Assessment has 2	3 Neutral assessment tests ( district for s 3 Neutral lents with disabili im roved the sub 3	4 A $ee$ NJ ASK 3-4, G students with di 4 A $ee$ ties believe the iect content the 4	5 EPA, HSPA) hav sabilities. 5 ee New Jersey ir child is learnin 5
than student erform 1 13. I believe the Ne im roved instruction 1 ee 14. In my view the Alternate Proficienc 1	nance. 2 Disa ee ew Jersey general a nal Disa ee parents of my stud Assessment has 2 Disa ee	3 Neutral assessment tests ( district for s 3 Neutral lents with disabili im roved the sub 3 Neutral	4 A ee NJ ASK 3-4, G students with di 4 A ee ties believe the iect content the 4 A ee	5 EPA, HSPA) hav sabilities. 5 ee New Jersey ir child is learnin 5
than student erform 1 13. I believe the Ne im roved instruction 1 ee 14. In my view the Alternate Proficienc 1	nance. 2 Disa ee ew Jersey general a nal Disa ee parents of my stud Assessment has 2 Disa ee	3 Neutral assessment tests ( district for s 3 Neutral lents with disabili im roved the sub 3 Neutral	4 A ee NJ ASK 3-4, G students with di 4 A ee ties believe the iect content the 4 A ee	5 EPA, HSPA) hav sabilities. 5 ea New Jersey ir child is learnin 5
than student erform 1 13. I believe the Ne im roved instruction 1 ee 14. In my view the Alternate Proficienc 1 15. I believe there a	nance. 2 Disa ee ew Jersey general a nal Disa ee parents of my stud Assessment has 2 Disa ee are alternative met	3 Neutral assessment tests ( district for s 3 Neutral lents with disabili im roved the sub 3 Neutral hods of assessment	4 A ee NJ ASK 3-4, G students with di 4 A ee ties believe the iect content the 4 A ee at other than the	5 EPA, HSPA) hav sabilities. 5 New Jersey ir child is learnin 5 current system
<ul> <li>than student erform</li> <li>1</li> <li>13. I believe the Neim roved instruction</li> <li>1</li> <li>ee</li> <li>14. In my view the Alternate Proficienc</li> <li>1</li> <li>15. I believe there a that would accurate!</li> </ul>	nance. 2 Disa ee ew Jersey general a nal Disa ee parents of my stud Assessment has 2 Disa ee are alternative metal assess student	3 Neutral assessment tests ( district for s 3 Neutral lents with disabili im roved the sub 3 Neutral hods of assessment ess for studen	4 A ee NJ ASK 3-4, G students with di 4 A ee ties believe the iect content the 4 A ee at other than the ts with disabilit	5 EPA, HSPA) hav sabilities. 5 ex New Jersey ir child is learnin 5 e current system ties.
<ul> <li>than student erform</li> <li>1</li> <li>13. I believe the Neim roved instruction</li> <li>1</li> <li>ee</li> <li>14. In my view the Alternate Proficience</li> <li>1</li> <li>15. I believe there a that would accuratel 1</li> </ul>	nance. 2 Disa ee w Jersey general a nal Disa ee parents of my stud Assessment has 2 Disa ee are alternative metal assess student 2	3 Neutral assessment tests ( district for s 3 Neutral and the sub 3 Neutral bods of assessment ass for studen 3 Neutral bods of assessment	$\begin{array}{c} 4\\ A & ee\\ NJ ASK 3-4, G\\ students with di 4\\ A & ee\\ ties believe the iect content the 4 A & ee\\ to ther than the ts with disability 4 $	5 EPA, HSPA) hav sabilities. 5 New Jersey ir child is learnin 5 e current system ties. 5

#### SCRIPT FOR FOCUS GROUP SESSION

# A Study of New Jersey's Assessment and Accountability System for Students with Disabilities and Its Impact on Special Education Services

# Julie Gardner

One of the most recent reform efforts to have a major impact on students with disabilities had been the No Child Left Behind Act and its mandate that all students participate in the state assessment process. New Jersey's assessment process is the NJ ASK 3-4, GEPA, HSPA and APA.

- 1. What evidence of change have you observed and experienced since the implementation of the New Jersey Assessment system relative to students with disabilities.
  - Probes:
  - a. Resources, class size
  - b. Instructional delivery options for students with disabilities
  - c. Participation rates in NJ Assessment (increase or decrease)
  - d. Curricular reform
- 2. What do you see as a positive result of the participation of students with disabilities in the assessment and accountability system.

Probes:

- a. Improved performance of students with disabilities on state/assessments
- b. Increased academic expectations for student with disabilities
- c. Increased networking between general and special educators
- d. Increased participation of special educators in training on standards and assessment
- e. Increased use of appropriate accommodation
- f. More students with disabilities accessing state academic content standards
- g. More students with disabilities in statewide general assessments
- 3. What do you see as a negative consequence of the participation of students with disabilities in the assessment and accountability system.

Probes:

- a. Few students with disabilities able to achieve "proficient" level on state general assessments/alternate proficiency assessment
- b. Including students with disabilities in the accountability system makes school and/or districts look less effective
- c. Too many students are identified to take the alternate assessment

An intended primary outcome of standards-based reform is improved student learning through the establishment of high standards for all students and assessment of student progress toward meeting these standards.

- 4. How does the NJ Assessment system do this for students with disabilities? Probes:
  - a. In relation to the NJ Core Curriculum Content Standards
- 5. What impact has the NJ Assessment program and the inclusion of students with disabilities had on child study teams and teachers in your districts? Probes:
  - a. Professional development opportunities
  - b. Knowledge of subject matter, curriculum, standards
  - c. Attitude/moral
- 6. What impact, if any, has the NJ Assessment program and the inclusion of students with disabilities had on the students.
  - Probes:
  - a. Retention and promotion
  - b. Attitude/moral
  - c. Knowledge of subject matter
- 7. What have the parents of your student said about the NJ assessment system Probes:
  - a. Impact on child's learning
  - b. Parent involvement
- 8. What impact has the inclusion of students with disabilities in the NJ Assessment process had on you district's accountability status?

# Probes:

- a. General education assessment results
- b. APA results
- 9. What approach does your district use to determine which students participate in the alternate assessment?
- 10. How are accommodations determined in your district?

- Probes: a. Accommodations used in day to day instruction and testing
  - b. Data collection on the accommodations that students use on state
    - assessments

11. What are some of the major obstacles to including students with disabilities in the district accountability system?

13. Describe any other methods of assessment you believe would meet the assessment requirements.

Probes:

- a. IEP Review
- b. Checklist

Describe emerging issues in relation to assessment or accountability in your district or within the state.